

Contents lists available at ScienceDirect

Heliyon

Heliyon

journal homepage: www.heliyon.com

Review Article

Quantum-classical mechanics as an alternative to quantum mechanics in molecular and chemical physics



Vladimir V. Egorov

Russian Academy of Sciences, FSRC "Crystallography and Photonics", Photochemistry Center, 7a Novatorov Street, Moscow, 119421, Russian Federation

ARTICLE INFO

Keywords: Theoretical physics Quantum mechanics Molecular quantum transitions Singularity Dozy chaos Quantum-classical mechanics Electron transfer Condensed matter Molecular physics Optics Applications

ABSTRACT

In quantum mechanics, the theory of quantum transitions is grounded on the convergence of a series of timedependent perturbation theory. In nuclear and atomic physics, this series converges because the dynamics of quantum transitions (quantum jumps) are absent by definition. In molecular and chemical physics, on the contrary, the dynamics of "quantum" transitions, being determined by the joint motion of a light electron (or electrons) and very heavy nuclei, are present by definition, and the series of time-dependent perturbation theory becomes singular. An exception is the dynamic problem for stationary states in the Born-Oppenheimer adiabatic approximation, when the electronic subsystem turns out to be "off" from the general dynamic process and therefore is not dynamically full-fledged: it only forms an electric potential in which the nuclei oscillate. Removing the aforementioned singularity can be accomplished in two ways. The first method was consisted of introducing an additional postulate in the form of the Franck-Condon principle into molecular quantum mechanics, in which the adiabatic approximation is used. The second method was proposed by the author and consisted of damping the singular dynamics of the joint motion of an electron and nuclei in the intermediate (transient) state of molecular "quantum" transitions by introducing chaos. This chaos arises only during molecular quantum transitions and is called dozy chaos. Formally, the damping is carried out by replacing an infinitely small imaginary addition in the spectral representation of the complete Green's function of the system with its finite quantity. The damping chaos (dozy chaos) leads to the continuity of the energy spectrum in the molecular transient state, which is a sign of classical mechanics. Meanwhile, the initial and final states of the molecule obey quantum mechanics in the adiabatic approximation. Molecular quantum mechanics, which takes into account the chaotic dynamics of the transient state of molecular "quantum" transitions, can be called quantum-classical (dozychaos) mechanics. The efficacy of the damping for the aforementioned singularity was previously shown by dozychaos mechanics of elementary electron transfers in condensed matter, which is the simplest case of dozy-chaos mechanics, and its applications to a whole number of problems, especially to the optical spectra in polymethine dyes and their aggregates. This paper provides a regular exposition of this dozy-chaos (quantum-classical) mechanics of the elementary electron transfers. The main results of its applications presented in the introduction are also described.

1. Introduction

1.1. A new theoretical approach to molecular quantum transitions and their applications

Quantum mechanics is one of the main, if not the most important, branches of modern theoretical physics. In the 20th century, atomic physics, nuclear physics, and solid state physics based on quantum mechanics created a fundamental basis for recent and modern technological progress. Therefore, even in a broad scientific environment, the possibilities of quantum mechanics are considered to be almost limitless. The applicability of quantum mechanics extends, for example, to the entire universe (see Refs. [1, 2, 3, 4]) and even the human brain and associated consciousness (see Refs. [5, 6, 7, 8, 9]). The word "quantum" is very

https://doi.org/10.1016/j.helivon.2019.e02579

Received 24 December 2018; Received in revised form 6 May 2019; Accepted 2 October 2019

2405-8440/© 2019 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Corresponding author. *E-mail address:* egorovphotonics@gmail.com.

fashionable in modern science (see Ref. [4]). In short, in the broad scientific community, there is a very popular view that quantum mechanics is the last and ultimate word about the essence of nature (see Ref. [9]). The current paper discusses the limits of the applicability of quantum mechanics based on an analysis of the internal contradictions that arise when it is applied unbiased to a wide range of problems in molecular and chemical physics.

Only a draft of the problem is outlined in the introduction. Later sections will include a detailed discussion on both the qualitative and quantitative levels. This section presents the main application results concerning the new approach proposed by the author in theory. This presentation differs from the generally accepted style, in which the theory is first presented and then its applications to the experiment are demonstrated. This is because most new results have already been published [10, 11, 12, 13, 14, 15, 16, 17, 18, 19] and reported at international conferences (see Refs. [20, 21, 22, 23, 24, 25, 26, 27, 28, 29, 30, 31, 32, 33, 34, 35, 36]). This article is based on publications and reports demonstrating the success of the new theory in applications to the experiment to discuss the new problem, that has arisen in physics, at a deeper physical level and provide a regular exposition of the theory in elementary electron transfers in condensed matter. There are two main reasons for simplifying this problem [11, 18]. The first involves approximating the electron's Green's function by it in the case of a free electron. The second is related to considering merely non-local vibrations of nuclei and disregarding their local vibrations.

A new theoretical approach to study of molecular quantum transitions and elementary electron transfers in condensed matter can be attributed accordingly to molecular and chemical physics. We call this new approach in theory dozy-chaos mechanics, or in another way, quantumclassical mechanics. The introduction of quantum-classical (dozy-chaos) mechanics into molecular and chemical physics has a forced character and is associated with the elimination of an essential singularity in the probabilities per unit time, or in other words, in the rate constants of quantum transitions in molecular systems. This singularity arises in quantum mechanics when it goes beyond the adiabatic approximation and follows from the incomparability of the masses of the electron (electrons) and nuclei and their joint motion in the process of molecular quantum transitions. The singularity is eliminated by changing the infinitesimal imaginary addition $i\gamma$ in the spectral representation of the complete molecular Green's function with its finite quantity [10, 11, 12]. The matching of the new theory with experiments shows that the modulus of this imaginary addition γ is much more than the quantum of nuclear oscillations $\hbar\omega$: $\gamma >> \hbar\omega$ [14, 28, 34]. The value of γ could be treated as the energy width of the electron-vibrational virtual levels of the transient state, which provides a multiple exchange of motion and energy between different vibrational modes of the nuclei and the electron in the transient state. However, because of the aforementioned excessively large value of γ , this exchange of motion and energy proves to be crash, and it causes chaos in the motion of both the nuclei and the electron that undergoes the transition. Since this chaos is absent in the initial and final states and present merely in the middle of molecular quantum transitions, it is called dozy chaos [14, 28, 33]. In chemical physics, the efficacy of the damping method for the aforementioned singularity was shown by the author with an example of the new (dozy-chaos) theory of elementary electron-charge transfers in condensed matter and its applications [10, 11, 12, 14, 15, 16, 17, 18, 19, 28, 29, 34, 35] to the optical spectra in polymethine dyes and their aggregates [37, 38, 39, 40, 41, 42, 43, 44] and its applications [20, 34, 45] to a number of other basic experimental data [46, 47].

The aforementioned quantum transitions in molecular and chemical physics (molecular quantum transitions), strictly speaking, are not quantum transitions, but quantum-classical transitions, since although their initial and final states are quantum, their transient states are of a classical nature. The quantum nature of the initial and final states is manifested in the fact that these states are often described by quantum mechanics in the adiabatic approximation and as a result of the reorganization of the nuclear subsystem, the structure of the final state of the molecular system differs markedly from the structure of its initial state. The latter fact is particularly evident for chemical reactions, which result in the formation of new molecules. The classical nature of the transient state of molecular systems is associated with the presence of chaos (dozy chaos) in the motion of electrons and nuclei involved in the quantum-classical transition, which leads to a continuous spectrum of their energies in this transient state [16, 18, 36, 48, 49]. The corresponding theory of quantum-classical transitions is called quantum-classical mechanics or dozy-chaos mechanics (see above). The aforementioned dozy-chaos theory of elementary electron-charge transfers in condensed matter is the simplest problem in quantum-classical mechanics¹ and will be described in detail in subsequent sections.

Quantum-classical mechanics provide insights into an entire series of the fundamental experimental results in chemistry, which in the past often resisted insight in the scope of the standard quantum mechanics of electron-nuclear motion. In the framework of quantum-classical mechanics, it is possible to explain, for example, experimental data on the shape of the optical bands of polymethine dyes and their aggregates in solutions, in which the quantum-classical transitions in their main optical chromophores can be approximated by elementary electron-chargetransfer processes in condensed matter [10, 11, 12, 14, 15, 18, 28]. Exciton effects arising in many cases as a result of the aggregation of molecules somewhat complicate the overall picture of elementary quantum-classical transitions in dye chromophores, but do not fundamentally change it [10, 11, 12, 14, 15, 18, 28].

To model the electronic structure of the basic optical chromophore in polymethine dyes, their polymethine chain, Dähne [50] put forward the concept of an ideal polymethine state, according to which there is a clearly extended distribution of the density of the π -electron charge along the quasi-linear polymethine chain. This density varies periodically along the chain and reallocates alternately along it during optical excitation (see Ref. [51]). Moreover, for the first excited state, the moment of the electronic transition is focused on the chain [51]. Therefore, the elementary electron-charge transfer along the chain can approximate the electronic transition to the first excited state [10, 11, 12, 14, 15, 18, 28]. The polymethine dyes to be discussed herein can be considered as the ideal polymethine state of Dähne.

Since for an ideal polymethine state, the total transfer of the alternating charge along the full chain consists of the acts of elementary transfer of a small amount of charge over a small distance between adjacent carbon atoms, the tunneling effects in such an electron-charge transfer are minor and the Gamow tunnel factor is close to unity [10, 11, 12, 14, 15, 18, 28].

The linearity and sufficient length of the basic optical chromophores of polymethine dyes and their aggregates, associated with the linearity and a sufficiently large length of their polymethine chain, lead to the fact that we can neglect the interaction of the electronic transition with the motion of the nuclei of the dyes themselves and take into account only its interaction with environmental nuclei [10, 11, 12, 14, 15, 18, 28].

The most important results are in theoretical optical spectra adjusted by the author to the fundamental experimental data on polymethine dye monomers (M) [12, 14, 15, 18, 28, 41, 42, 43] (see Figs. 1 and 2), dimers (D) [15, 29, 42, 43], H and H* aggregates [16, 18, 29, 42, 43] (see Fig. 2), and J aggregates [10, 11, 12, 14, 17, 18, 28, 29, 37, 38, 39, 40, 42, 43] (see Fig. 2) and also theoretical spectra adjusted to the generally known data on the M–D [15, 29, 42, 43] and M–J aggregate [10, 11, 12, 14, 28, 42, 43] concentration equilibriums.

Among these important results, a special place is occupied by a theoretical explanation of the nature of the well-known narrow J band (see Fig. 2), which is determined by the dynamic pumping of an

¹ The simplicity is associated herein with approximating the electron's Green's function by it in the case of a free electron and considering merely non-local vibrations of nuclei and disregarding their local vibrations (see above) [11, 18].



Fig. 1. Experimental [41, 43] (*a*) and theoretical [12] (*b*) monomer's optical absorption spectra dependent on the length of the polymethine chain (thiapolymethinecyanine in methanol at room temperature; ε is the extinction coefficient) [15, 18, 19]. (Original citation) — Reproduced by permission of The Royal Society of Chemistry.



Fig. 2. Theoretical optical absorption spectra [18, 19, 29, 35] (b) in thiapolymethinecyanines adjusted to the fundamental experimental data (a) on polymethine dye monomers (M), dimers (D), H-, H*-, and J-aggregates [43].

electronic transition in the chromophore of J aggregates (a tetramer with a brickwork-type structure) with weakly chaotic, or in other words, highly organized motion of nuclei in the environment (the case of weak dozy chaos) [10, 11, 12, 14, 18, 28]. The role of exciton effects in the shaping of the J band is insignificant. Conversely, the nature of the narrow H* band (see Fig. 2) is elucidated by the existence of dozy-chaos dynamics as well as a reasonably great exciton dynamics, their strong interference interaction [16, 18, 29]. A competitive interference of the chaotic dynamics of nuclear reorganization and exciton dynamics in the chromophore of the H * aggregate, which is the dimer (H * dimer), leads to a "pumping chaos" from the peak of the optical band of the H * dimer into its wing, making the peak even narrower and the wing still wider. This effect is associated with an abnormally strong exciton interaction in the H* dimer, which is a consequence of the specific structure of the H* monomers that make up it.

The H * monomers are cyclic bis-thiacarbocyanines that have an abnormally large area of a planar optical chromophore (see Ref. [42]). Therefore, there is a strong exciton interaction of the H * monomers in the H * dimer, which leads to the narrowing of the peak of its optical band as a result of "pumping chaos" from the peak into the wing of the band [16, 18, 29].

The striking results of quantum-classical mechanics are also the results of explaining the resonance behavior of the shape of the optical band of the polymethine dye monomer as a result of a change in the length of its polymethine chain (see Fig. 1) [12, 15, 18, 19, 41, 43] and a change in sign of the relative intensity of the two bands closest to the resonance band (see Fig. 1, n = 3) as a result of a change in the polarity of the solvent [17, 18, 44]. Other dozy-chaos explanations can be also found in Ref. [17]. Dozy-chaos (quantum-classical) mechanics of "quantum" transitions in the discussed organic substance appears fairly sophisticated, but this substance in its facility in dynamical nature rates in quantum-classical mechanics like a hydrogen atom in structural nature in quantum mechanics [33b].

All of the results of the optical spectra match generally weak dozy chaos ($\gamma << E$, where *E* is the so-called reorganization energy introduced in Section 2.4.1). Strong dozy chaos ($\gamma \ge E$) leads to the elucidation of the important patterns in the reactions of proton transfer [45, 47] and comparatively fresh temperature-dependent effects on electron transfers in Langmuir-Blodgett films [20, 46]. In the case of strong dozy chaos, the dynamics of quantum-classical transitions become weakly dependent on dozy chaos, and the electronic component of the complete electron-nuclear amplitude of transitions can be fitted by the Gamow tunnel exponential dependent on the transient phonon environment. This elementary method permit us to evade the consideration of the imaginary additive $i\gamma$ in the spectral representation of the complete Green's function and to word the physical nature of the transient state not in the concept of dozy chaos, but in the concept of a large number of tunnel and over-barrier energy states providing the "quantum" transition of an

elementary charged particle. This method was worked out [52] long before the development of quantum-classical mechanics [11, 12, 14, 15, 18, 28] (see the subsequent sections), and now we can say that the concept of a large number of tunnel and over-barrier states is the simplest version of the concept of dozy chaos.

Grounded on this method [52], in 1990 a theoretical description of the basic experimental patterns in the Brönsted relationships [47] for the reactions of proton transfer (acid-base catalysis) was given [45]. The Brönsted relationships was found by Brönsted and Pedersen in 1924 (see Ref. [47]). The theory in Ref. [52] is immediately appropriate to the explanation of electron transfers. To explain the reactions of transfers of heavy charged particles (proton transfers), the result of thermic fluctuations of the potential barrier transparence must be considered because of fluctuations of the barrier width. In contrast to the elementary proton transfer, the electron transfer process is insensitive to small fluctuations of the barrier width due to the large size of the electronic wave function in the initial and final states. The analytical formulas for the proton-transfer rate constants are obtained. In acid catalysis, the Brönsted relationship is $\lg K^{(\text{acid})} = \alpha \lg K_{\text{eq}} + a$, where $K^{(\text{acid})}$ is the rate constant, K_{eq} is the equilibrium constant, and α and a are constants. In base catalysis, the Brönsted relationship is $\lg K^{(\text{base})} = \beta \lg K_{\text{eq}} + b$. The theoretically obtained Brönsted coefficients α and β for direct and inverse reactions meet the generally known empirical equality $\alpha + \beta = 1$. The experimentally ascertained large extent of linearity of the Brönsted relationships is elucidated by the concept of a large number of tunnel and over-barrier states, and by the phonon frequencies dispersion and the barrier width fluctuations. The kinetic isotope effect (proton, deuteron or triton transfer) computed using our theory is in agreement with experimental data (see Ref. [45]). In some cases (see Ref. [45]) the regularities of acid-base reactions are coupled not with the proton transfers, but with the electron transfers. The criteria for selecting one or another mechanism of reactions are formulated in Ref. [45]. By considering the generally known similarity of the relationships of Brönsted and Tafel's law in electrochemistry, the theory of acid-base catalysis can be adapted to electrochemical reactions [45].

Ref. [20] provides an explanation for the temperature dependence of the activation energy $E_a = E_0(1 - T/T_0)$ (where $T_0 \approx 350$ K) for electron transfers at high temperatures in Langmuir-Blodgett films, which is discovered by Naito and Miura [46], as well as an interpretation of the entire set of these authors' experimental data in a wide diapason of temperatures. The activation energy E_a reduces and proves to be negative as temperature T grows because the domain of tunnel states expands on the energy scale and nears the apex of the electron potential barrier, and the quantity of over-barrier states grows.

Section 2.1 considers fundamental questions of the theory of molecular quantum transitions.

2. Main text

2.1. Nature of molecular quantum transitions and the concept of dozy chaos

At the end of the 19th century, most physicists believed that the physical picture of the world was complete or nearing completion. This physical picture of the world was based on classical mechanics. However, the theoretical description of the equilibrium radiation of an absolutely black body, the problem of the so-called ultraviolet catastrophe, and the constancy of the velocity of light and its independence from the motion of the reference frame remained to be elucidated. The theory of relativity and quantum mechanics emerged as a result.

Quantum mechanics soon led to the development of atomic and nuclear physics. Quantum mechanics was also the basis for molecular physics, which rapidly developed in two directions. The first was associated with the theory of the structure of molecules and solids, which was based on the Born-Oppenheimer adiabatic approximation [53]. The second was associated with molecular optical spectroscopy based on the Franck-Condon principle [54, 55, 56, 57]. The adiabatic approximation, in which the motion of light electrons quickly adapts to the slow motion of heavy nuclei and which often perfectly describes the stationary (ground) state of molecular systems, is in full compliance with the principles of quantum mechanics. The Franck-Condon principle assumes that during optical excitation of a molecule, a light electron makes a fast quantum jump (transition) to an excited state, and then the whole configuration of heavy nuclei in the molecular system slowly adapts to the new distribution of the electron charge in the excited state. This assumption leads in many cases to a good agreement between theoretical and experimental results of molecular optical spectra.

Unlike in atomic spectroscopy, where there is no physical meaning, with a few exceptions,² in molecular spectroscopy, the dynamics of a quantum transition are of considerable significance and are already determined by the Franck-Condon principle.³ Although the Franck-Condon dynamics of the quantum transition cause a number of *a priori* physical objections,⁴ due to the good agreement of the theory with experiments, for almost 100 years it has been an "unshakable" physical postulate within molecular quantum mechanics.

Thus, the generally accepted theory of molecular quantum transitions has a long history based on the well-known classical works of Born and Oppenheimer [53] and Franck and Condon [54, 55, 56, 57], which appeared immediately after the development of the foundations of quantum mechanics. Born and Oppenheimer were the first to solve the quantum-mechanical problem for a system of coupled electrons and nuclei in the simplest case for molecules [53], offering an effective and successful physical and mathematical technique, later called adiabatic approximation. Adiabatic approximation originated from adiabatic invariants, which were previously formulated by Ehrenfest [60]. The effectiveness of adiabatic approximation results from the Born-Oppenheimer theory of a significant difference in the masses of electrons and nuclei, which enables the separation of the slow motion of very heavy nuclei from the rapid motion of light electrons by neglecting the small nonadiabaticity operator in the Schrödinger equation to solve quantum-mechanical problems. The Born-Oppenheimer theory [53] is the basis of modern solid-state physics and quantum chemistry.

A completely different picture forms in connection with the works of Franck and Condon [54, 55, 56, 57], the basis of the so-called Franck--Condon principle. As previously mentioned, due to the significant difference in the masses of electrons and nuclei, the dynamics of molecular quantum transitions consists of two stages. In the first stage, under the influence of an external perturbation, a light electron nearly instantaneously passes into an excited electronic quantum state. In the second stage, a so-called reorganization of the nuclear subsystem of the molecule occurs, in which the spatial configuration of the heavy nuclei slowly adapts to the new electron-charge distribution. However, such a picture raises a serious objection. Namely, due to the significant inertia of the nuclei, they will not follow the movement of the electron, but simply return the electron to its original state just as rapidly [14, 28, 33b]. In other words, proceeding from these general considerations, the probability (per unit time) of a molecular quantum transition must be zero. However, if the heavy nuclei suddenly begin to move after a light electron already in the excited state, then due to the substantial inertia of the nuclei, they cannot stop in the excited electronic state [14, 28, 33b]. In

² One exception is the Lamb shift in atomic spectroscopy [58, 59].

³ Formally, there are no quantum transition dynamics in the Franck-Condon principle. This principle simply assumes that, as the initial and final states enter the matrix element of the quantum transition, the wave functions are considered in the adiabatic approximation. However, this assumption can be interpreted physically as previously indicated.

⁴ For example, why do nuclei with substantial inertia follow the new distribution of the electron charge rather than returning the excited electron to the initial state? See details below.

other words, the probability (per unit time) of a molecular quantum transition must be infinite. Thus, these simple and general qualitative considerations demonstrate that there is a singularity in the rates of the molecular quantum transitions. At one time, these thoughts were either not discussed or considered, apparently because of the success of the Franck-Condon principle in its practical applications, which resulted in the development of modern molecular optical spectroscopy. Nevertheless, the question of the validity of the Franck-Condon principle remained unchanged in theory. In this regard, the Franck-Condon principle should not be treated as a physical principle, but as an efficient hypothesis that does not have a thorough theoretical justification.

In addition to the previously mentioned qualitative objections, there is no formal quantitative substantiation for the Franck-Condon principle in theory. Proceeding from the first principles of quantum mechanics, it is necessary to quantitatively solve the problem of the dynamics of the transient state in molecular quantum transitions, at least in some simple cases. Moreover, such a solution, particularly substantiating the Franck-Condon principle, should clearly be demonstrated in many diverse applications. However, such a solution has not been obtained to date.

Returning to the qualitative physical consideration: "Under what condition is a light electron able to move nuclei with substantial inertia and thereby allow the reorganization of the nuclear subsystem in molecular quantum transitions?" [14, 28, 33b]. The electron should abandon attempts to move the nuclei "alone" but somehow provoke the same nuclei to move. How does this occur if the nuclei in the molecule have at least vibrational motion and at all temperatures up to absolute zero (zero-point vibrations)? To shift the equilibrium positions of oscillating nuclei in space, their vibrational motions must be transformed, at least in part, into the translational motion of their equilibrium positions during the molecular quantum transition. This is easy to implement if the motions of the nuclei are chaotic during the molecular quantum transition. If the vibrations of the nuclei are chaotic, an electron could easily control their motion during "quantum" transition. As shown by the formal theory, which will be discussed later, electrons that bind nuclei (atoms) to molecules cause chaotic motion in nuclei in a transient state. In other words, a light electron, through the creation of chaos, organizes the motion of very heavy nuclei during molecular "quantum" transition, resulting in this transition (that is, it occurs at a non-zero and non-infinite rate as previously mentioned). This is the essence of the self-organization of molecular quantum transitions. This occurs through the creation of chaos by an electron in a transient molecular state. This is called dozy chaos as discussed in Section 1. Dozy chaos is the joint effect of the collective chaotic motion of electrons and nuclei, and their chaotic electromagnetic interactions in the transient state of molecules experiencing quantum transitions. Dozy chaos is a universal physical phenomenon, because electrons and nuclei are universal [14, 15, 16, 18, 28, 32, 33, 34, 35, 36, 48, 49].

One can cite some a priori objection to the previously mentioned qualitative reasoning, from which it follows that there is a singularity (that is, zero or infinity) in the rates of molecular transitions. This objection is that the nuclei, even because of their substantial inertia, cannot return an excited light electron to its original ground state since the electron is already in a well-defined quantum state from which it cannot escape because of the presence of an energy gap in the electronic energy according to quantum mechanics. Therefore, in the excited state, the electron is forced to slowly adjust the nuclear subsystem to its new charge distribution. This constitutes the essence of the Franck-Condon principle. However, our a priori objection and the Franck-Condon principle itself are entirely based on the *a priori* assumption that quantum mechanics works in this case. As previously mentioned, this assumption has no formal evidence. Moreover, qualitative considerations about the continuous energy spectrum in the transient state, which appears due to the electron provocation of chaos in the vibrational motion of the nuclei to control their motion in this state (as previously mentioned), indicate that in reality there is no gap in the electronic energy, and consequently, quantum mechanics ceases to work in a transient dynamic molecular state. In other words, the excited electron in principle cannot be

"hooked" into the quantum state, which follows from the standard solution to Schrödinger's equation, and therefore it quickly "rolls" over the continuous spectrum of energy to the ground state, as discussed in our initial qualitative reasoning.

Considering quantum mechanics as applied to molecular quantum transitions, according to the Franck-Condon principle, in optical excitation the electron transition occurs at the turning points of the oscillating nuclei, that is, at the moment they stop. This "reasonable assumption" in quantum mechanics⁵ makes it possible to "switch off" the nuclear subsystem at this time from a dynamic analysis, just as in the framework of quantum mechanics the electronic subsystem is "switched off" from a full-fledged dynamic analysis in a justified procedure to isolate the nonadiabaticity operator in Schrödinger's equation and neglecting it in the adiabatic approximation (see Section 2.3).

The Franck-Condon principle essentially follows from two physical facts. The first assumes (in general, unreasonable, as previously noted) that quantum mechanics works in the field of molecular quantum transitions. The second assumes that, over the course of their classical oscillations in an oscillatory potential well, the nuclei spend a majority of the time near the turning points (see footnote 5). Therefore, the quantum transition (see the first fact) has an overwhelming probability of occurring at these points, and consequently, it will be vertical or close to vertical. Thus, the same oscillator behaves in the same elementary molecular way both as a classical system (near the bottom of an oscillatory potential well for nuclei, where the probability of a quantum transition is negligible) and as a quantum system (near the turning points of the nuclei, where the probability of a quantum transition is maximal). Therefore, the oscillator is essentially a quantum-classical system (see also Section 2.5), and the Franck-Condon principle is essentially one of the most important prerequisites for the creation of quantum-classical mechanics. During the formulation of the Franck-Condon principle [54, 55, 56, 57], the prospect of creating quantum-classical mechanics was not noticed, apparently because of the strongest "psychological confrontation" between quantum and classical mechanics.

The time of the classical behavior of the oscillator near its bottom (or, equivalently, far from the turning points of the nuclei) is much shorter than the period of the oscillations of the nuclei. This correlates with the fact in quantum-classical mechanics that the time of molecular "quantum" transition or, equivalently, the time of reorganization of the nuclear subsystem \hbar/E (*E* is the energy of the reorganization of the nuclear subsystem as mentioned in Section 1), when, due to chaos in the transient state, the electron-nuclear system leads like the classical system (see above), is also much shorter than the nuclear oscillation period $\frac{\hbar}{E} << \Im$, where $E = (1 \div 0.1)$ eV and $\Im \cong 10^{-13}$ s ($(10^{-15} \div 10^{-14})$ s $<< 10^{-13}$ s).

The dynamics of the transient state in molecular quantum transitions can be approached from the other side, namely, from the characters of the optical spectra observed in the experiment. The discrete line spectrum in atoms is a direct indication of the existence of quantum jumps in atoms. The nature of molecular spectra differs from that of atomic spectra: they are striped or often even continuous, as for example in the case of polymethine dyes and their aggregates (see Section 1). The continuity of the spectra, observed for polymethine dyes and their aggregates, indicates that the transition from one quantum state to another is not a quantum jump but a transition through a continuous spectrum of electron-vibrational states.⁶ (The continuity of the spectra is also associated with the dispersion of the phonon frequencies; see Section 2.9).

In strong dozy chaos ($\gamma \geq E$), quantum-classical mechanics are compatible with the Born–Oppenheimer [53] and Franck–Condon [54,

⁵ The usual reasoning is: "It is obvious that the nuclei spend most of their time near the turning points. Therefore, the probability of the electron and 'intended' vertical transition near these points will be maximal."

⁶ Since the energy of the rotational quantum is much less than the energy of the vibrational quantum, for simplicity, hereinafter, we take into account only the vibrational system of degrees of freedom and ignore the rotational one.

55, 56, 57] quantum mechanics with high precision [11, 12, 14, 15, 16, 18, 28, 34] (see Sections 2.4.9, 2.5, and 2.9). It turns out that in this particular case, the rates of "quantum" transitions, as in standard molecular quantum mechanics, depend only on the initial and final states, since due to strong dozy chaos they do not depend on the dynamics of the transient state. In other words, dozy chaos often "leaves no trace" in molecular physics, which is why it took so long to discover it. In these often practiced cases, standard molecular quantum mechanics gives results that are consistent with the experiment, despite the fact that they are based on erroneous concepts [11, 12, 14, 15, 16, 18, 28, 34]. In other words, the Franck-Condon principle is an effective simulator of strong dozy chaos, both in the problem of elementary electron transfers in condensed matter, and, presumably, in all other cases of molecular "quantum" transitions (see details in Section 2.5).

Extremely strong dozy chaos ($\gamma >> E$) leads to standard molecular quantum mechanics, but with abnormally low rates of "quantum" transitions. Weak dozy chaos ($\gamma << E$) causes a strong dynamic self-organization of molecular "quantum" transitions and hence their high rates [11, 12, 14, 15, 16, 18, 28, 34] (see Section 1).

2.2. Divergence of a series of time-dependent perturbation theory in quantum mechanics for molecular quantum transitions

Turning to formal techniques of quantum mechanics, we can precisely indicate the point at which quantum mechanics applied to molecular quantum transitions ceases to work. As is well known, the theory of quantum transitions is based on the time-dependent perturbation theory (TDPT) in quantum mechanics (see *e.g.* Ref. [61]). The time-dependent Schrödinger equation is solved by the standard scheme of the perturbation theory [61]:

$$i\hbar\frac{\partial\Psi}{\partial t} = \mathbf{H}(t)\Psi, \quad \mathbf{H}(t) = \mathbf{H}_0 + V(t), \quad V(t) = \begin{cases} W(t), & \text{if } 0 \le t \le \tau, \\ 0, & \text{if } t < 0, t > \tau \end{cases}.$$
(1)

The series for the transition amplitude A_{fl} is as follows

$$A_{fl}(t) = \sum_{n=0}^{\infty} \frac{\langle f | I_n(t) | l \rangle}{n!},\tag{2}$$

where

$$I_{n}(t) = \left(\frac{1}{i\hbar}\right)^{n} P \int_{0}^{t} dt_{1} \int_{0}^{t} dt_{2} \dots \int_{0}^{t} dt_{n} \tilde{W}(t_{1}) \tilde{W}(t_{2}) \dots \tilde{W}(t_{n}),$$
(3)

P is the Dyson's chronological operator and

$$\tilde{W}(t) = e^{\frac{i}{\hbar}H_0 t} W(t) e^{-\frac{i}{\hbar}H_0 t}$$
(4)

is the perturbation operator in the interaction representation. The series of TDPT converges in atomic and nuclear physics, as in the matrix elements of transitions $\langle f | I_n(t) | l \rangle$ in the amplitude (2), due to quantum jumps, the dynamics of quantum transitions is not contained by definition, and for many problems it suffices to confine oneself to the first order of TDPT. The probability of a quantum transition from state $| l \rangle$ to state $| f \rangle$ during the time of the perturbation τ is determined by the formula ("Fermi's golden rule", see Ref. [61])

$$\mathfrak{T}_{fl}(\tau) = \left| A_{fl}^{(1)}(\tau) \right|^2 = \frac{1}{\hbar^2} \left| \int_0^\tau \langle f | \tilde{W}(t) | l \rangle e^{i\omega_{fl}t} dt \right|^2, \tag{5}$$

where $\hbar \omega_{fl} = E_f - E_l$, E_f and E_l are the energies of stationary states.

The series of TDPT converges also in the case of molecular quantum transitions (and electronic transitions in condensed matter), which occur in accordance with the Franck-Condon principle [54, 55, 56, 57], that is, when the states $|l\rangle$ and $|f\rangle$ in the amplitude (2) are taken in the

Born-Oppenheimer adiabatic approximation [53]. In other words, the physical picture of quantum transitions in molecular physics is usually represented as quantum jumps, which are similar to quantum jumps in atomic and nuclear physics. (A small parameter that ensures the convergence of the series is a small perturbation time τ , see Eq. (1).) The resulting theoretical (Franck-Condon [54, 55, 56, 57]) transition probabilities (per unit time) that follow from Eq. (5) often explain the experimental data well, for example, in molecular spectroscopy (see e.g. Refs. [62, 63, 64]). However, in chemistry, for example, there are many experimental data (see e.g. Section 1), which can not be explained within the adiabatic approximation. This forces us to go beyond the adiabatic approximation. But such an exit beyond the adiabatic approximation leads to divergence of the series of TDPT in quantum mechanics, which is associated with the incomparability of the masses of electrons and nuclei, jointly participating in the dynamics of the transition, which, unlike to atomic and nuclear transitions, is contained here by definition. To verify this fact, the divergence of the series of TDPT is enough to demonstrate by any one example. As such an example, it is convenient to consider the simplest problem in the theory of quantum transitions, which goes beyond the adiabatic approximation. Such a simple problem, as it turned out, is the problem of the theoretical description of the elementary electron transfers in condensed matter [11, 18]. As previously mentioned (Section 1 and footnote 1), the simplicity of this problem is related to the sufficiency of taking into account only the non-local phonons in the matter and the possibility of approximating the electron Green's function by the propagator (see Sections 2.4.7 and 2.4.4 below) [11, 18].

Before proceeding the formulation of this simplest theory, that is, a new theory of elementary electron transfers in condensed matter, the author believes that for its better understanding it is important to remind the reader of the essence of the Born-Oppenheimer adiabatic theory.

2.3. The Born-Oppenheimer adiabatic approximation⁷

The standard base for a general study of electron-vibrational interactions is the adiabatic theory [53, 65, 66, 67], which uses the only universal small parameter of the molecule — the Born-Oppenheimer parameter $\kappa = (\mu/M)^{1/4}$, where μ is the electron mass and M is the typical mass of the nucleus.

The Hamiltonian \hat{H} of a molecule is written as the sum of the kinetic energy of electrons \hat{T}_e and nuclei \hat{T}_N and the total potential energy U (q,p) of the molecule:

$$\widehat{H} = \widehat{T}_e + \widehat{T}_N + U(q, p), \tag{6}$$

where *q* and *p* are the set of electronic and nuclear coordinates. In the adiabatic approximation the wave function $\Psi(q,p)$ of the molecule is searched in the form

$$\Psi(q,p) = \psi(q,p) \phi(p). \tag{7}$$

Substituting Eq. (7) into the stationary Schrödinger equation

$$\widehat{H}\Psi(q,p) = E\Psi(q,p),\tag{8}$$

where the Hamiltonian \hat{H} is given by Eq. (6), we obtain

$$\phi \widehat{T}_e \psi + \widehat{T}_N \psi \phi + U \psi \phi = E \psi \phi. \tag{9}$$

We formally introduce the operator \hat{L} defined from the equation:

⁷ In this section and throughout this article, we assign the names of the pioneers of the theory of molecules, namely, the names of Born and Oppenheimer, to the adiabatic approximation, although between the adiabatic theory of Born and Oppenheimer [53] and the modern interpretation of the adiabatic approximation [65, 66, 67], which we cite here, there are some differences, which are discussed, for example, in Refs. [66, 67].

V.V. Egorov

$$\widehat{L}\psi\phi = \widehat{T}_N\psi\phi - \psi\widehat{T}_N\phi.$$
(10)

Substituting $\hat{T}_N \psi \phi$ from Eq. (10) into Eq. (9) and dividing both sides of the equation by $\psi \phi$, we get

$$\frac{1}{\psi}\widehat{T}_{e}\psi + U + \frac{1}{\psi\phi}\widehat{L}\psi\phi = E - \frac{1}{\phi}\widehat{T}_{N}\phi.$$
(11)

Let us mark the right part of this equation, which depends solely on the nuclear coordinates p, through V(p); then from Eq. (11) and this notation we obtain

$$\left(\widehat{T}_{e} + U + \frac{1}{\phi}\widehat{L}\phi\right)\psi = V\psi, \qquad (12)$$

$$(\widehat{T}_N + V)\phi = E\phi.$$
(13)

Eqs. (12) and (13) are identical to the original Schrödinger Eq. (8). It is only rewritten for the new formally introduced functions $\psi(q,p)$ and $\phi(p)$ in Eq. (7).

The adiabatic approximation agrees with the disregard for the term $(1/\phi)\hat{L}\phi$ in Eq. (12), which is called the nonadiabaticity operator. Hence the system of adiabatic equations has the form

$$[T_e + U(q,p)] \psi_f = V_f(p) \psi_f,$$
(14)

$$\left[\widehat{T}_N + V_f(p)\right]\phi_{fn} = E_{fn}\phi_{fn}.$$
(15)

The solution of the electron Schrödinger Eq. (14) gives a system of electron wave functions and energy levels that depend on the nuclear coordinates p as parameters. The electron energy $V_f(p)$ acts as a potential function for the motion of the nuclei (see Eq. (15)). This function is often called the electron term. The concept of electron term or potential energy surface (PES) plays a key role not only in the theory of molecules and quantum chemistry as a whole, but also in the modern theory of elementary chemical reactions and elementary processes of electron-charge transfers. If the use of PESs in the former case is perfectly justified, then in the latter case, when it comes to quantum transitions in the PES intersection area (in the transient region), the Born-Oppenheimer (BO) adiabatic approximation does not work at all. This can be seen if we write out the corrections to the BO approximation in perturbation theory with respect to the nonadiabaticity operator:

$$V_{f}^{(1)} = \left\langle \psi_{f}^{(0)} | \hat{T}_{N} | \psi_{f}^{(0)} \right\rangle,$$
(16)

$$V_{f}^{(2)} = \sum_{f' \neq f} \frac{\left| \left\langle \psi_{f'}^{(0)} \middle| \widehat{T}_{N} \middle| \psi_{f}^{(0)} \right\rangle - \sum_{p} \frac{\hbar^{2}}{M} \frac{1}{\phi} \frac{\partial \phi}{\partial p} \left\langle \psi_{f'}^{(0)} \middle| \frac{\partial \psi_{f}^{(0)}}{\partial p} \right\rangle \right|^{2}}{V_{f}^{(0)} - V_{f'}^{(0)}}$$
(17)

and

$$\psi_{f}^{(1)} = \sum_{f' \neq f} \frac{\left\langle \psi_{f'}^{(0)} \middle| \widehat{T}_{N} \middle| \psi_{f}^{(0)} \right\rangle - \sum_{p} \frac{\hbar^{2}}{d}}{V_{f}^{(0)}} \frac{\frac{\partial \phi}{\partial p}}{\sqrt{\psi_{f'}^{(0)}}} \left\langle \psi_{f'}^{(0)} \middle| \frac{\partial \psi_{f}^{(0)}}{\partial p} \right\rangle}{V_{f}^{(0)} - V_{f'}^{(0)}} \cdot \psi_{f'}^{(0)}.$$
(18)

The BO approximation is violated for the electron energy in the second order of perturbation theory and for the electron wave function in the first order of perturbation theory. It can be seen that the nonadiabatic corrections $V_f^{(2)}$ from Eq. (17) and $\psi_f^{(1)}$ from Eq. (18) are not minor in the case when the differences of the electronic terms $V_f^{(0)} - V_{f'}^{(0)}$, which

depend on the nuclear coordinates p and which enter into the denominators, turn out to be small.

Thus, the PESs, being, as a rule, dynamic (adiabatic) invariants for the initial and final states of the electron-nuclear system,⁸ are not dynamic invariants for the transient state. Therefore, the problem of finding dynamical invariants for the transient state arises. We will solve this problem in the simplest case of "chemical transformations": in the case of extended electron-vibrational transitions or, in other words, in the case of elementary electron transfers in condensed matter.

2.4. Quantum-classical mechanics of elementary electron transfers in condensed matter

2.4.1. Adiabatic approximation for elementary electron transfers

In comparison with the Hamiltonian in the standard theory of manyphonon transitions (see Ref. [72]), in the theory of elementary electron transfers the Hamiltonian is complicated merely by a complementary electron potential well $U_2(\mathbf{q} - \mathbf{L})$ set apart from the original well $U_1(\mathbf{q})$ by the distance $L \equiv |\mathbf{L}|$ [11]:

$$\mathbf{H} = -\frac{\hbar^2}{2\mu} \Delta_{\mathbf{q}} + U_1(\mathbf{q}) + U_2(\mathbf{q} - \mathbf{L}) + \sum_{\iota} U_{\iota}(\mathbf{q}) \ p_{\iota} + \frac{1}{2} \sum_{\iota} \hbar \omega_{\iota} \left(p_{\iota}^2 - \frac{\partial^2}{\partial p_{\iota}^2} \right),$$
(19)

where μ is the effective mass of an electron, **q** is the electron's radius vector, p_i are the normal phonon coordinates (real), ω_i are the eigenfrequencies of normal oscillations, and i is the phonon index; $\sum_{i} U_i(\mathbf{q}) p_i$ is

the electron-phonon coupling expression.

In the adiabatic approximation, the solution of the Schrödinger equation

$$H \Psi = E_H \Psi$$
(20)

for the system "electron + environment" is sought in the form

$$\Psi(\mathbf{q},p) = \psi(\mathbf{q},p) \, \Phi(p) \tag{21}$$

(compare with Eq. (7) in Section 2.3), where the electron ψ -function satisfies the Schrödinger equation

$$\left[-\frac{\hbar^2}{2\mu}\Delta_{\mathbf{q}}+U_1(\mathbf{q})+U_2(\mathbf{q}-\mathbf{L})+\sum_{i}U_i(\mathbf{q})p_i\right]\psi(\mathbf{q},p)=E(p)\psi(\mathbf{q},p).$$
 (22)

If we neglect the nonadiabaticity operator (compare with Eq. (10) in Section 2.3)

$$\widehat{L}\Psi \equiv \widehat{T}_{p}\psi\Phi - \psi\widehat{T}_{p}\Phi = -\sum_{i}\hbar\omega_{i}\left(\frac{\partial\psi}{\partial p_{i}}\frac{\partial\Phi}{\partial p_{i}} + \frac{\Phi}{2}\frac{\partial^{2}\psi}{\partial p_{i}^{2}}\right)$$
(23)

$$\left(\widehat{T}_{p} = -\frac{1}{2}\sum_{\iota} \hbar \omega_{\iota} \frac{\partial^{2}}{\partial p_{\iota}^{2}}\right) \text{ in Eq. (20), then}$$

$$\left[E(p) + \frac{1}{2}\sum_{\iota} \hbar \omega_{\iota} \left(p_{\iota}^{2} - \frac{\partial^{2}}{\partial p_{\iota}^{2}}\right)\right] \Phi(p) = E_{\mathrm{H}}^{\mathrm{BO}} \Phi(p), \qquad (24)$$

where $E_{\rm H}^{\rm BO}$ is the approximate eigenvalue of the total Hamiltonian (19). The index BO in Eq. (24) and beyond indicates that the corresponding quantity is accepted in the Born-Oppenheimer adiabatic approximation.

The electronic Eq. (22) is solved by perturbation theory with the perturbation operator

$$\tilde{U} \equiv \sum_{i} U_{i}(\mathbf{q})(p_{i} - \tilde{p}_{i}),$$
(25)

similar to the perturbation operator of Pekar [73, 74]. It will be shown below that \tilde{p}_i are the values of the normal phonon coordinates for which

⁸ The adiabatic approximation is often disturbed even in the ground state, for example, due to vibronic interactions [66, 68, 69] and in non-rigid molecular systems [70, 71].

the adiabatic potential in Eq. (24) has a minimum.

In the zero order of the perturbation theory with respect to the operator \tilde{U} , for the electron Eq. (22) we have

$$\left[-\frac{\hbar^2}{2\mu}\Delta_{\mathbf{q}}+U_1(\mathbf{q})+U_2(\mathbf{q}-\mathbf{L})+\sum_{\iota}U_{\iota}(\mathbf{q})\;\tilde{p}_{\iota}\right]\psi_s(\mathbf{q})=E_s^0\psi_s(\mathbf{q}),\qquad(26)$$

where the index *s* numbers the states of the discrete and continuous spectra. It is assumed that the interaction

$$\sum_{\iota} U_\iota(\mathbf{q}) \; ilde{p}_\iota$$

at the donor 1 and at the acceptor 2 is the same, that is, by definition

$$\tilde{p}_{l2} = -\tilde{p}_{l1} \equiv -\tilde{p}_{l} \, (\tilde{p}_{l} < 0) \tag{27}$$

and

$$U_{l2}(\mathbf{q} - \mathbf{L}) = -U_{l1}(\mathbf{q}) \equiv -U_{l}(\mathbf{q}).$$
(28)

In this wise

$$\sum_{i} U_{i2}(\mathbf{q} - \mathbf{L}) \tilde{p}_{i2} = \sum_{i} [-U_{i1}(\mathbf{q})] [-\tilde{p}_{i1}] = \sum_{i} U_{i1}(\mathbf{q}) \tilde{p}_{i1} \equiv \sum_{i} U_{i}(\mathbf{q}) \tilde{p}_{i}.$$

The first order correction to the electron energy $E_s^0(\tilde{q})$ is

$$\int d\mathbf{q} \boldsymbol{\psi}_{s}^{*}(\mathbf{q}) \tilde{U}(\mathbf{q}, p) \boldsymbol{\psi}(\mathbf{q}) = \int d\mathbf{q} \boldsymbol{\psi}_{s}^{*}(\mathbf{q}) \left[\sum_{i} U_{i}(\mathbf{q})(p_{i} - \tilde{p}_{i}) \right] \boldsymbol{\psi}_{s}(\mathbf{q})$$

$$= \sum_{i} U_{is}(p_{i} - \tilde{p}_{i}),$$
(29)

where

$$U_{is} \equiv \int d\mathbf{q} U_i(\mathbf{q}) |\psi_s(\mathbf{q})|^2.$$
(30)

Thus, in the first order of perturbation theory with respect to the operator \tilde{U} , for the electron energy we have

$$E_s(p) = E_s^0 + \tilde{U}_s(p), \quad \tilde{U}_s(p) \equiv \sum_i U_{is}(p_i - \tilde{p}_i).$$
(31)

So, in the presence of an electron (on a donor or on an acceptor), the operator of nuclear energy in the adiabatic approximation has the form:

$$E_{s}(p) + \frac{1}{2} \sum_{i} \hbar \omega_{i} \left(p_{i}^{2} - \frac{\partial^{2}}{\partial p_{i}^{2}} \right) = E_{s}^{0} + \sum_{i} U_{is}(p_{i} - \tilde{p}_{i}) + \frac{1}{2} \sum_{i} \hbar \omega_{i} \left(p_{i}^{2} - \frac{\partial^{2}}{\partial p_{i}^{2}} \right).$$
(32)

The role of the potential energy of nuclei is played by the function

$$F_{s}(p) = E_{s}^{0} + \sum_{i} U_{is}(p_{i} - \tilde{p}_{i}) + \frac{1}{2} \sum_{i} \hbar \omega_{i} p_{i}^{2}.$$
(33)

We find its minimum:

$$\frac{dF_s}{dp_i} = U_{is} + \hbar\omega_i p_i, \quad \frac{dF_s}{dp_i}|_{p_i = \bar{p}_i} = 0, \quad \frac{d^2F_s}{dp_i^2} = \hbar\omega_i > 0;$$

$$\bar{p}_i = -\frac{U_{is}}{\hbar\omega_i}, \quad (34)$$

where U_{ls} are given by Eq. (30). Suppose that $U_l(\mathbf{q}) \approx \text{constant}$. We choose this constant in the following form: $\text{constant} \equiv -\hbar \omega_l \tilde{p}_l$. Then, taking

$$\int d\mathbf{q} |\psi_s(\mathbf{q})|^2 = 1,$$

we obtain

$$\overline{p}_i = \widetilde{p}_i, \tag{35}$$

that is, in the aforementioned assumption, \tilde{p}_i are the values of the normal phonon coordinates at which the adiabatic potential of Eq. (24) for the nuclear wave function $\Phi(p)$ has a minimum. Thus, the function $U_i(\mathbf{q})$ is assumed to be equal to its average over the electron state in the zero order of perturbation theory with respect to the operator \tilde{U} .

The minimum value of the potential energy of the nuclei in Eq. (33) is

$$F_{s}(p)|_{p=\bar{p}} = E_{s}^{0} + \frac{1}{2} \sum_{i} \hbar \omega_{i} \tilde{p}_{i}^{2} \equiv E_{s}^{0} + E \equiv J_{s},$$
(36)

where

$$E \equiv \frac{1}{2} \sum_{i} \hbar \omega_{i} \tilde{p}_{i}^{2}$$
(37)

is the energy of reorganization of the nuclear vibrations due to the presence of an electron on the donor or the acceptor. The potential energy of the nuclei is written in terms of the quantities J_s in the form

$$F_s(p) = J_s + \frac{1}{2} \sum_i \hbar \omega_i (p_i - \tilde{p}_i)^2.$$
(38)

In the two-level approximation $J_s \equiv -J_{1,2} < 0$: in the initial state the electron is on the donor 1, and in the final state — on the acceptor 2. The corresponding potential energy surfaces of the nuclei $F_{1,2}(p)$ are the two paraboloids of dimension M, where M is the number of vibrational degrees of freedom of the crystal, the vertices of which, according to Eq. (27), are displaced relative to each other by the amount $2|\tilde{p}|$. The operator of the total energy of nuclei (Eq. (32)) has the form

$$-J_{1,2} + \frac{1}{2} \sum_{i} \hbar \omega_{i} \left[\left(p_{i} - \tilde{p}_{i} \right)^{2} - \frac{\partial^{2}}{\partial p_{i}^{2}} \right].$$
(39)

Its eigenvalues are equal to

$$E_{\rm H}^{\rm BO} = -J_{1,2} + \sum_{i} \hbar \omega_i \left(m_{i1,2} + \frac{1}{2} \right), \tag{40}$$

and its eigenfunctions are equal to

$$\begin{split} \Phi_{1,2\dots,m_{l1,2}\dots}(p-\tilde{p}) &= \prod_{i} \varphi_{m_{l1,2}}(p_{i}-\tilde{p}_{i}), \\ \varphi_{m_{l1,2}}(p_{i}-\tilde{p}_{i}) &= A_{m_{l1,2}}e^{-\frac{V_{2}(p_{i}+\tilde{p}_{i})^{2}}{H_{m_{l1,2}}(p_{i}+\tilde{p}_{i})}}H_{m_{l1,2}}(p_{i}+\tilde{p}_{i}). \end{split}$$
(41)

Here $H_m(p) = (-1)^m e^{p^2} d^m e^{-p^2} / dp^m$ are the Hermite polynomials of degree m (m = 0, 1, 2, ...), A_m is the normalizing factor.

Thus, for the wave function of the "electron + environment" system in the Born-Oppenheimer adiabatic approximation (Eq. (21)) and according to Eq. (26), we have

$$\Psi_{1,2}^{\text{BO}}(\mathbf{q},p) = \psi_{1,2}(\mathbf{q})\Phi_{1,2\dots m_{l+2}\dots}(p-\tilde{p}).$$
(42)

An adiabatic approximation of the type Eq. (42), in which the electron wave function ψ does not depend on the nuclear coordinates p, is often called the rough adiabatic approximation.

2.4.2. Technique of Green's functions

In accordance with our aim to discover "successful" dynamic invariants for the transient state [12, 13], which would be alternative to the Born-Oppenheimer adiabatic invariants — potential energy surfaces, we search for the solution of the Schrödinger Eq. (20) for the "electron + environment" system by Green's function technique. At the initial stage of constructing this solution, the identical transformations of Eq. (20) in the

Green's function method can be considered as alternatives to identical transformations when the non-adiabaticity operator (23) is singled out in the Born-Oppenheimer method (see Section 2.4.1): if the purpose of the latter is to separate the slow motion nuclei from the rapid motion of an electron, the purpose of the former is to maximally preserve the inter-connection of electronic and nuclear movements.

So, to the Hamiltonian of the "electron + environment" system of some general form H = H(q, p), we add and subtract some, while arbitrary, operator $\tilde{U} = \tilde{U}(q, p)$:

$$\mathbf{H} \equiv \mathbf{H} - \tilde{U} + \tilde{U}. \tag{43}$$

Then the Schrödinger Eq. (20) can be rewritten in the form

$$(\mathbf{H} - U - E_{\mathbf{H}}) \Psi = -U \Psi.$$
(44)

We consider the right-hand side in Eq. (44) as an inhomogeneity. The corresponding homogeneous equation has the form

$$(\mathbf{H} - \tilde{U} - E_{\mathbf{H}}) \ \tilde{\Psi} = 0. \tag{45}$$

Let us introduce the Green's function $G_{\rm H}$ and G:

$$(\mathbf{H} - E_{\mathbf{H}})G_{\mathbf{H}}(\mathbf{q}, \mathbf{q}'; p, p'; E_{\mathbf{H}}) = -\Delta(\mathbf{q} - \mathbf{q}')\Delta(p - p'),$$
(46)

$$(\mathbf{H} - \tilde{U} - E_{\mathbf{H}})G(\mathbf{q}, \mathbf{q}'; p, p'; E_{\mathbf{H}}) = -\Delta(\mathbf{q} - \mathbf{q}')\Delta(p - p').$$
(47)

They are connected by the Dyson integral equation

$$G_{\rm H}(\mathbf{q}, \mathbf{q}'; p, p'; E_{\rm H}) = G(\mathbf{q}, \mathbf{q}'; p, p'; E_{\rm H}) + \iint d\mathbf{q}_{\rm I} dp_{\rm I} G(\mathbf{q}, \mathbf{q}_{\rm I}; p, p_{\rm I}; E_{\rm H}) \tilde{U}(\mathbf{q}_{\rm I}, p_{\rm I}) G_{\rm H}(\mathbf{q}_{\rm I}, \mathbf{q}'; p_{\rm I}, p'; E_{\rm H})$$
(48)

(this can be easily verified by acting the operator $(H - \tilde{U} - E_H)$ on the left and right sides of the equation). In a symbolic way, the Dyson equation (Eq. (48)) has the form

$$G_{\rm H} = G + G\tilde{U}G_{\rm H}.\tag{49}$$

The solution of Eq. (49) is found by successive approximations

$$G_{\rm H} = G + G \tilde{U} G + G \tilde{U} G \tilde{U} G + \dots$$
(50)

The general solution of Eq. (44) for the wave function $\Psi(\mathbf{r}, q)$ of the "electron + environment" system has the form

$$\Psi(\mathbf{q},p) = \tilde{\Psi}(\mathbf{q},p) + \iint d\mathbf{q}' dp' G(\mathbf{q},\mathbf{q}';p,p';E_{\mathrm{H}}) \tilde{U}(\mathbf{q}',p') \Psi(\mathbf{q}',p').$$
(51)

This integral equation is the Lippmann-Schwinger equation. It is identical to the original Schrödinger Eq. (20) or (44) (this can be easily verified by acting the operator $(H - \tilde{U} - E_H)$ on the left and right sides of the equation and taking into account Eq. (45)). We rewrite the Lippmann-Schwinger equation (Eq. (51)) in the symbolic form

$$\Psi = \tilde{\Psi} + G\tilde{U}\Psi. \tag{52}$$

We find its solution by successive approximations

$$\begin{aligned} \Psi &= \tilde{\Psi} + G \,\tilde{U} \,\tilde{\Psi} + G \,\tilde{U} \,G \,\tilde{U} \,\tilde{\Psi} + G \,\tilde{U} \,G \,\tilde{U} \,G \,\tilde{U} \,\tilde{\Psi} + \dots \\ &= \tilde{\Psi} + (G + G \,\tilde{U} \,G + G \,\tilde{U} \,G \,\tilde{U} \,G + \dots) \,\tilde{U} \,\tilde{\Psi}. \end{aligned} \tag{53}$$

Taking into account the solution of the Dyson equation for $G_{\rm H}$ (see Eq. (50)), we obtain

$$\Psi = \tilde{\Psi} + G_{\rm H} \ \tilde{U} \ \tilde{\Psi}. \tag{54}$$

This wave function Ψ is a formal solution of the original Schrödinger Eq. (20) or (44). Our next problem is, by a proper choice of the operator \tilde{U} in the solution (54), to separate the state of elementary transfer of the electron, which we are interested in, from nonphysical solutions corresponding to other types of electron and nuclear motions. To do this, it is

necessary to specify the Hamiltonian H = H(q, p) of the "electron + environment" system, which figures while in some general form in our Green's function method.

In the simplest case, we have Hamiltonian (19), which was already used by us earlier in considering the Born-Oppenheimer adiabatic approximation. We choose the operator \tilde{U} , that figures in our method of Green's functions, in the form of the operator (25), which we used earlier (Section 2.4.1) in the Born-Oppenheimer method:

$$\tilde{U} = \sum_{i} U_i(\mathbf{q})(p_i - \tilde{p}_i).$$
(25a)

Here, in contrast to Eq. (25), \tilde{p}_i are some, so far arbitrary, constants. Then

$$\mathbf{H} - \tilde{U} = -\frac{\hbar^2}{2\mu} \Delta_{\mathbf{q}} + U_1(\mathbf{q}) + U_2(\mathbf{q} - \mathbf{L}) + \sum_{\iota} U_{\iota}(\mathbf{q}) \ \tilde{p}_{\iota} + \frac{1}{2} \sum_{\iota} \hbar \omega_{\iota} \left(p_{\iota}^2 - \frac{\partial^2}{\partial p_{\iota}^2} \right)$$
(55)

and the solution of the homogeneous Eq. (45),

$$\tilde{\Psi}(\mathbf{q},p) = \psi(\mathbf{q})\Phi_0(p),\tag{56}$$

corresponds to the lack of interaction of the electron with the vibrations of the nuclei. In other words, the nuclei vibrate as if there were no electron in the condensed medium, and the motion of the electron occurs as if the nuclei were immovably fixed in their equilibrium positions. Thus, the solution $\tilde{\Psi}(\mathbf{q}, p)$ does not correspond to either the electron-transfer state or many-phonon transitions in general. Therefore, it must be excluded from the general solution (54) as describing an unphysical state. Consequently, the state of electron transfer is determined by the solution [79].

$$\Psi = G_{\rm H} \ \tilde{U} \ \tilde{\Psi}. \tag{57}$$

2.4.3. Electron-phonon wave functions of the initial and final states. On the convergence of a series of time-dependent perturbation theory in quantum-classical mechanics

In the solution (57), however, the fact of the "start" of the electron from the donor 1 is not yet reflected. This fact is taken by substituting the wave function in the adiabatic approximation (42) into account in the right-hand side of Eq. (57), which corresponds to the presence of interaction of the electron with the vibrations of the nuclei, instead of the wave function (56), which corresponds to the lack of this interaction. Formally, this reduces only to a shift in the normal phonon coordinates by an amount \tilde{p} in the phonon part of the wave function. As a result, the constants \tilde{p}_i introduced above (Eq. (25*a*)) acquire the physical meaning of the shifts of the normal phonon coordinates, which correspond to the shifts in the equilibrium positions of the nuclei, caused by the presence of an electron in the medium. In other words, the quantities \tilde{p}_i in the method of Green's functions have the same physical meaning as in the Born-Oppenheimer method. Thus, the state of electron transfer at its "start" from the donor 1 is described by the wave function [79].

$$\Psi_1 = G_{\rm H} \tilde{U} \Psi_1^{\rm BO}. \tag{58}$$

We recall that in the solution $G_{\rm H}$ (see Eq. (50)) of the Dyson Eq. (49) the Green's function *G* is determined by the Hamiltonian (55) with the separating variables **q** and *p*. So, in the Green's function $G_{\rm H}$ it remains to find only the value of the total energy $E_{\rm H}$ of the system. This energy is taken in the adiabatic approximation corresponding to the wave function $\Psi_1^{\rm BO}$ (see Eqs. (24) and (40, 41, 42)). As a result, we obtain the wave function (58), where the Green's function $G_{\rm H} = G_{\rm H}(E_{\rm H} = E_{\rm H}^{\rm BO})$ is expressed in terms of $G = G(E_{\rm H} = E_{\rm H}^{\rm BO}, i\gamma \rightarrow 0)$ as a series (50).

The infinitesimal imaginary additive $i\gamma$ specified in *G* is ordinarily written in the spectral representation of the Green's function in order to escape zero in the energy denominator. In our case, the spectral repre-

sentation of the Green's function has the form:

$$G \equiv G(\mathbf{q}, \mathbf{q}'; p, p'; \mathbf{H} - \tilde{U}) = \sum_{s} \frac{\Psi_{s}(\mathbf{q}, p) \Psi_{s}^{*}(\mathbf{q}', p')}{E_{\mathbf{H}}^{\mathsf{BO}} - (E_{s}(p) - \tilde{U}(p)) - i\gamma},$$
(59)

where $\Psi_s(\mathbf{q}, p)$ are the eigenfunctions of the Hamiltonian (55) and $E_{\rm H}^{\rm BO} = E_s(p)$ (see Section 2.4.1); the energy denominator vanishes when $\tilde{U}_s(p=\tilde{p})=0$ ($\tilde{U}_s(p)\equiv \sum U_{is}(p_i-\tilde{p}_i)$, see Eq. (31)) and $\gamma = 0$. We,

however, ascribe a finite value to the value of γ , as a result of which γ acquires the physical meaning of the degree of chaos in the reorganization of the nuclei of the medium, provoked by the motion of the electron from the donor towards the acceptor.⁹ Thus, for the elementary electron transfers, along with the energy of reorganization $E \equiv \frac{1}{2} \sum_{i} \hbar \omega_{i} \tilde{p}_{i}^{2}$ (see Eq. (37)) of the nuclei of a medium, a dissipation energy γ is introduced into the theory, which characterizes the degree of chaos in the motion of nuclei in a transient state. The implementation of the finite quantity γ makes it possible to escape the singularity in the rates of elementary electron transfers, which is formally associated with the vanishing of the energy denominator of the complete Green's function of the system at $\gamma = 0$, and which is physically associated with the incomparability of the masses of the electron and nuclei of the environment,¹⁰ and it indicates physically insertion, to the transient state, of a process for the transformation, at first, of a portion of the vibrational motion of the nuclei into their translational motion, and after that, in the course of completing the "quantum" transition, of the emerged translational motion of the nuclei back into their vibrational motion. Since the chaos in the electronnuclear movement is absent in the initial and final states and occurs merely in the transient state, it is called *dozy chaos*, and the appropriate dissipation energy γ can be called the *dozy-chaos energy* [14, 28].

Along with the objective of damping the singular electron-nuclear movement, our procedure of introducing a finite magnitude of γ has extra important objective: it allows us to ensure the parameter of smallness of the problem [79].

$$\tilde{U} \ G \sim \frac{\tilde{U}}{\gamma} \sim \frac{\hbar \omega_{\iota}}{\gamma} << 1.$$
(60)

Hence

(

$$G >> G \tilde{U} G >> G \tilde{U} G >> G \tilde{U} G \tilde{U} G >> ...,$$
(61)

and in obedience to the series (50), the Green's function $G_{\rm H}$ is as follows

$$G_{\rm H} \approx G.$$
 (62)

Thus, in accordance with Eq. (58), we finally have a solution

describing the state of electron transfers:

$$\Psi_1 \approx G \; \tilde{U} \; \Psi_1^{\rm BO},\tag{63}$$

where $G = G(E_{\rm H} = E_{\rm H}^{\rm BO}; i\gamma, \gamma > > \hbar\omega_i)$.

Starting from relationship (60), it is not difficult to understand that the wave function of the system for an electron localized on the acceptor 2, in order to avoid nullity in the transition amplitude

$$A_{12} = \langle \Psi_2(\mathbf{q} - \mathbf{L}, p) | \mathbf{W} | \Psi_1(\mathbf{q}, p) \rangle, \tag{64}$$

should be taken no longer in the form of Eq. (63), but in the adiabatic approximation (Eq. (42)): $\Psi_2 = \Psi_2^{\text{BO}}$.¹¹

The series for the transition rate constant, which corresponds to the Green's function series (Eq. (50)), according to relationship (60) has a small parameter

$$\left(\overline{m}_{1}\hbar\omega_{i}/\gamma\right)^{2} < <1,\tag{65}$$

where \overline{m}_1 is the distribution function of Planck.¹² Therefore, in our problem for $k_BT > \hbar \omega_t/2$, as a small parameter, we have

$$\left(k_{\rm B}T/\gamma\right)^2 < <1. \tag{66}$$

So, according to Eq. (60), the parameter for the convergence of the Dyson series for the Green's function $G_{\rm H}$ (Eq. (50)) is

$$\frac{\hbar\omega}{\gamma} << 1 \tag{67}$$

(for simplicity, we assume the vibrational frequency ω =constant $\equiv \omega$). Therefore, the same parameter (Eq. (67)) ensures the convergence of the TDPT series (see Section 2.2) for the amplitude of the quantum transitions $A_{fl}(t)$ (see Eq. (2)), in which the initial (*l*) and final (*f*) states are determined by the regular wave functions previously obtained: the initial state — by the dynamic wave function according to Eq. (63), and the final state — by the wave function taken in the adiabatic approximation (Eq. (42)).

2.4.4. The simplest electron-phonon Green's function

In Eq. (63), the Green's function $G(\mathbf{q}, \mathbf{q}'; \mathbf{L}; p, p'; E_{\mathrm{H}}^{\mathrm{BO}})$ is given by the expression following from its spectral representation (59) and the form of the Hamiltonian (55) with the separating variables \mathbf{q} and p:

$$G(\mathbf{q}, \mathbf{q}'; \mathbf{L}; p, p'; E_{\mathrm{H}}^{\mathrm{BO}}) = \sum_{\dots, m_{t}, \dots} G_{\varepsilon}(\mathbf{q}, \mathbf{q}'; \mathbf{L}; \tilde{p}; E_{\mathrm{H}}^{\mathrm{BO}} - \varepsilon_{\dots, m_{t}, \dots}) \Phi_{0\dots, m_{t}, \dots}(p) \Phi_{0\dots, m_{t}, \dots}(p'),$$
(68)

where

$$G_{e}(\mathbf{q},\mathbf{q}';\mathbf{L};\tilde{p};E_{\mathrm{H}}^{\mathrm{BO}}-\varepsilon_{\dots m_{i}\dots})=\sum_{s}\frac{\psi_{s}(\mathbf{q};\mathbf{L};\tilde{p})\psi_{s}^{*}(\mathbf{q}';\mathbf{L};\tilde{p})}{(E_{\mathrm{H}}^{\mathrm{BO}}-\varepsilon_{\dots m_{i}\dots})-E_{s}(\mathbf{L};\tilde{p})-i\gamma}$$
(69)

is the Green's function of the extended electron motion associated with the virtual motion of the nuclei. It follows from the form of the Hamiltonian (55) with separating variables **q** and *p* that the electronic wave functions $\psi_s(\mathbf{q}; \mathbf{L}; \hat{p})$ in Eq. (69) are the eigenfunctions of Eq. (26),¹³ and the phonon wave functions $\Phi_{0...n_{t...}}(p)$ in Eq. (68) are given by Eq. (41) for

⁹ Usually, the quantity γ specifies the rule for bypassing singular points in contour integration. However, in our theory of charge transfers [11], some mathematical technique [52, 75] (see Section 2.4.6) manages to perform an exact summation over the intermediate states that are associated with γ . This "releases γ from former duties" and lets it to gain a new status. The author's mathematical technique [52, 75] is a substantial development of the generating polynomial (generating function) technique of Krivoglaz and Pekar [76, 77], and the Krivoglaz-Pekar technique [76, 77] in its turn is analogous to computing the thermodynamic quantities by the Darwin-Fowler technique [72, 78], known from statistical physics.

¹⁰ We note that, in analogy with the standard theory of many-phonon transitions [72], our original Hamiltonian (Eq. (19)) depends on the electronic and normal phonon coordinates of the system, and in solving this problem on its basis it is not clear how exactly, because of the incomparability of the masses of the electron and nuclei, a singularity is "formed" in the observed physical quantities, the emergence of which follows from the above analysis of the problem on a qualitative physical level (see Section 2.1). This singularity manifests itself formally in the observed physical quantities, either as γ tends to zero ($\gamma \rightarrow 0$) (see Fig. 4 in Section 2.6) or for $\gamma \equiv 0$ in the whole of the original theory.

¹¹ For optical transitions, the perturbation in Eq. (64) is accepted in the regular long-wave approximation: W = p, where p is the momentum operator of the electron (see Ref. [61]).

¹² The factor \overline{m}_1 appears from considering the equilibrium distribution of the population for m_1 initial states of phonons.

¹³ That is, they coincide with the eigenfunctions of the electronic Schrödinger's Eq. (22), which are obtained in the zero order of the adiabatic perturbation theory of Pekar [72, 73] with respect to the operator (25).

 $\tilde{p} = 0$ and replacing the indices of the initial and final states 1, 2... $m_{t1,2}$... by the index of intermediate states 0... m_t The phonon energies in Eq. (69) are

$$\varepsilon_{\dots m_{l}\dots} \equiv \sum_{i} \hbar \omega_{i} \left(m_{i} + \frac{1}{2} \right)$$
(70)

(compare with Eq. (40) in Section 2.4.1).

The general expression for the electron Green's function (Eq. (69)) has the form (see Ref. [80])

$$G_e = G^{(\text{free})} - G^{(\text{free})} \widehat{T} G^{(\text{free})}, \tag{71}$$

where

$$G^{\text{(free)}} \equiv G^{\text{(free)}}(\mathbf{q}, \mathbf{q}'; k) = \frac{\mu}{2\pi\hbar^2} \frac{\exp(\pm ik|\mathbf{q} - \mathbf{q}'|)}{|\mathbf{q} - \mathbf{q}'|}$$
(72)

is the propagator or the Green's function of the free motion of an electron with energy $\hbar^2 k^2/2\mu$, and \hat{T} denotes the *T*-operator (scattering operator). In our case, in the denominator of G_e (see Eq. (69)), according to Eqs. (40) and (70), we have the electron energy

$$E_{\rm H}^{\rm BO} - \varepsilon_{\dots m_1 \dots} - i\gamma = -(J_1 + \hbar \omega_1 + i\gamma), \tag{73}$$

where

$$G(\mathbf{q}, \mathbf{q}'; p, p') = \sum_{\dots, m_{l}, \dots} G_{e}[\mathbf{q}, \mathbf{q}'; \alpha(\dots, m_{l}, \dots)] \Phi_{0\dots, m_{l}, \dots}(p) \Phi_{0\dots, m_{l}, \dots}(p'),$$
(79)

where the electron Green's function $G_e[\mathbf{q}, \mathbf{q}'; \alpha(..., m_i ...)]$ is given by Eq. (77), and the phonon wave functions $\Phi_{0...m_{l...}}(p)$ are given, as previously indicated, by Eq. (41) for $\tilde{p} = 0$ and replacing the indices of the initial and final states 1, 2... $m_{i1,2}$... by the index of intermediate states $0...m_{i...}$

2.4.5. The simplest Green's function of elementary electron transfers

As a result of neglecting the *T*-operator term in Eq. (71), the simplest electron-phonon Green's function (Eq. (79)) does not depend on the distance *L* between the donor and the acceptor of the electron charge. The dependence on the distance *L* arises after integration over coordinates **q** of the electron in the transfer amplitude (Eq. (64)). Therefore, the definition of the simplest Green's function for the elementary electron-charge transfers must include this integration.

Substituting the expression (58) for the wave function $\Psi_1(\mathbf{q}, p)$ written in detail in terms of the Green's function according to Eqs. (68) and (69) and expression (42) for the wave function $\Psi_2(\mathbf{q} - \mathbf{L}, p)$ in the adiabatic approximation into the amplitude given by Eq. (64), we obtain

$$A_{12} = \sum_{l} \sum_{\dots,m_{l}\dots} A_{l} \left[J_{1} + \sum_{i} \hbar \omega_{i} (m_{i} - m_{i1}); L; \tilde{p} \right] \frac{o(m_{l}, m_{l1})}{r(m_{l}, m_{l1})} \prod_{i} r(m_{i2}, m_{i}) r(m_{i}, m_{i1})$$
(80)

where (taking into account Eq. (35))

$$A_{l}\left[J_{1}+\sum_{i}\hbar\omega_{i}(m_{i}-m_{i1});L;\tilde{p}\right]=-\hbar\omega_{l}\tilde{p}_{l1}\iint d\mathbf{q}d\mathbf{q}'G_{e}[\mathbf{q},\mathbf{q}';\alpha(\dots m_{i} \dots)]\psi_{2}^{*}(\mathbf{q})|\mathbf{q}-\mathbf{L}|\psi_{1}(\mathbf{q}'),\tag{81}$$

$$\omega_1 \equiv \sum \omega_t (m_t - m_{t1}). \tag{74}$$

From here

$$k = \pm i [2\mu (J_1 + \hbar \omega_1 + i\gamma)]^{1/2} / \hbar.$$
(75)

In the problem under consideration, the properties of the *T*-operator are determined by the form of the electron potential wells $U_1(\mathbf{q})$ and $U_2(\mathbf{q} - \mathbf{L})$. We choose these potentials in the simplest form, namely, in the form of the zero-radius Fermi potentials [81, 82]. It is clear that in this case, for not too small distances $|\mathbf{L}| \equiv L$, which we have for the transfer of the electronic charge, the term $G^{(\text{free})} \hat{T}G^{(\text{free})}$ in Eq. (71) does a significant part just in such small surroundings of points $\mathbf{q} = 0$ and $\mathbf{q} = \mathbf{L}$ (see Ref. [83]), which give a small contribution to the transfer rate constant [11, 79]. Therefore, for G_e (see Eq. (69)) we have

$$G_e \cong G^{\text{(free)}}.\tag{76}$$

Taking into account Eqs. (72) and (75), we obtain

$$G_e(\mathbf{q}, \mathbf{q}'; \alpha) = \frac{\mu}{2\pi\hbar^2} \frac{\exp(-\alpha|\mathbf{q} - \mathbf{q}'|)}{|\mathbf{q} - \mathbf{q}'|},\tag{77}$$

where

$$\alpha \equiv \alpha(\dots m_i \dots) \equiv \alpha(\omega_1) = \left[2\mu(J_1 + \hbar\omega_1 + i\gamma)\right]^{1/2} / \hbar$$
(78)

(ω_1 is given by the formula (74)). Thus, substituting G_e from Eq. (77) into Eq. (68), for the simplest electron-phonon Green's function we have

$$\prod_{i} r(m_{i}, m_{i1,2}) \equiv \prod_{i} \int \varphi_{m_{i}}(p_{i}) \varphi_{m_{i1,2}}(p_{i} - \tilde{p}_{i}) dp_{i}$$
$$= \int \Phi_{0...m_{i...}}(p) \Phi_{1,2...m_{i1,2}...}(p - \tilde{p}) dp,$$
(82)

$$o(m_{l}, m_{l1}) \equiv -\int \varphi_{m_{l1}}(p_{l}' - \tilde{p}_{l})(p_{l}' - \tilde{p}_{l})\varphi_{m_{l}}(p_{l}')dp_{l}', \qquad (83)$$

where, for their part, $\varphi_{...}(...)$ are the wave functions of the linear harmonic oscillator (Eq. (41) for $\tilde{p} = 0$ and/or $\tilde{p} \neq 0$). The factor $|\mathbf{q} - \mathbf{L}|$ in the integrand of Eq. (81) comes from the perturbation operator W in the transition amplitude (Eq. (64)), which is accepted in the long-wave approximation (see Section 2.4.3).

Using the zero-radius approximation [81, 82] for the electron potential wells $U_1(\mathbf{q})$ and $U_2(\mathbf{q}-\mathbf{L})$ (see Section 2.4.4), for the wave functions $\psi_1(\mathbf{q}')$ and $\psi_2^*(\mathbf{q})$ in Eq. (81) we have

$$\psi_1(\mathbf{q}') = \left(\frac{\alpha_1}{2\pi}\right)^{1/2} \frac{\exp(-\alpha_1 |\mathbf{q}'|)}{|\mathbf{q}'|},\tag{84}$$

$$\psi_2^*(\mathbf{q}) = \left(\frac{\alpha_2}{2\pi}\right)^{1/2} \frac{\exp(-\alpha_2 |\mathbf{q} - \mathbf{L}|)}{|\mathbf{q} - \mathbf{L}|},\tag{85}$$

where

$$\alpha_{1,2} = \frac{[2\mu(J_{1,2} - E)]^{1/2}}{\hbar},$$
(86)

and E is the energy of reorganization of the vibrations of the nuclei (see

Eq. (37)), $J_1 > J_2 > E$ by definition. Substituting $G_e[\mathbf{q}, \mathbf{q}'; \alpha(..., m_t, ...)]$ from Eq. (77) and $\psi_1(\mathbf{q}')$ and $\psi_2^*(\mathbf{q})$ from Eqs. (84) and (85) into formula (81) and computing the integral $\int d\mathbf{q}'...$, we get

$$A_{l} = -\frac{\mu(\alpha_{1}\alpha_{2})^{1/2}}{(2\pi\hbar)^{2}}\hbar\omega_{l}\tilde{p}_{l1}\int I(|\mathbf{q}|)\exp(-\alpha_{2}|\mathbf{q}-\mathbf{L}|)d\mathbf{q},$$
(87)

where

$$I(|\mathbf{q}|) = \frac{4\pi[\exp(-\alpha|\mathbf{q}|) - \exp(-\alpha_1|\mathbf{q}|)]}{(\alpha_1^2 - \alpha^2)|\mathbf{q}|}.$$
(88)

As a result, after computing the integral $\int ...d\mathbf{q}$ in Eq. (87), we have [11, 79].

$$A_l = c_{12}\omega_l \tilde{p}_{l1} G^{\mathrm{E}},\tag{89}$$

where

$$c_{12} = -\frac{8\mu\alpha_2(\alpha_1\alpha_2)^{1/2}}{\hbar},$$
(90)

the Green's function of the elementary electron transfers is

$$G^{\rm E} = G^{\rm E}(\alpha, L) = \frac{1}{L} \left[\frac{\exp(-\alpha L) + f(\alpha, L)}{(\alpha_1^2 - \alpha^2)(\alpha_2^2 - \alpha^2)^2} \right],\tag{91}$$

$$f(\alpha, L) \equiv \left\{ \left[1 + \frac{(\alpha_2^2 - \alpha_1^2)L}{2\alpha_2} \right] \exp(-\alpha_2 L) - \exp(-\alpha_1 L) \right\} \frac{(\alpha_2^2 - \alpha^2)^2}{(\alpha_2^2 - \alpha_1^2)^2} - \left[\frac{L}{2\alpha_2} \exp(-\alpha_2 L) \right] (\alpha_2^2 - \alpha^2) - \exp(-\alpha_2 L), (92) \right\}$$
(92)

and the function $\alpha = \alpha(\omega_1)$ is given by formula (78). Thus, the Green's function of the electron-charge transfers $G^E = G^E(\omega_1, L)$ to the distance *L* is given by Eqs. (91) and (92), in which the function $\alpha = \alpha(\omega_1)$ is given by Eq. (78). The values of $\hbar\omega_1$, on which α depends, are the energy levels of the electron in the transient state, which arise as a result of virtual

$$P_{l}(m_{i2}, m_{i1}; \omega_{1}) \equiv \sum_{\dots, m_{l}, \dots} \frac{o(m_{l}, m_{l1})}{r(m_{l}, m_{l1})} \prod_{l} r(m_{l2}, m_{l}) r(m_{l}, m_{l1}) \Delta \left[\omega_{1} - \sum_{l} \omega_{l}(m_{l} - m_{l1}) \right]$$

movements of the electron and nuclei.

2.4.6. The general expression for the rate constant of elementary electron photo-transfers. The technique of generating functions

Taking into account the quasi-discreteness of the phonon energy spectrum in a condensed matter having a bounded volume, we introduce an energy unit $\hbar \overline{\omega}$ so small that all the phonon energies $\hbar \omega_i$ are expressed by integers [72, 76, 77]. The heat energy of elementary electron-charge transfers is then given by an integer $\Delta \equiv \hbar \omega_{12} > 0$.¹⁴ Then, proceeding according to the rules of quantum mechanics, within the "Golden Fermi rule", the rate constant of the elementary electron-charge photo-transfers (optical extinction) ε for the ω_{12} -phonon act can be formulated in the following way [72, 73, 74]:

$$c = \frac{4\pi^2 q^2 M_{\rm A} \Omega}{3\hbar c n_{\rm ref}} K$$
⁽⁹³⁾

where *q* is the quantity of electron charge transferred in a quantumclassical transition, Ω is the circular frequency of the absorbed photon, M_A is the Avogadro constant, *c* is the speed of light in vacuum, n_{ref} is the refractive index, and the quantity $K = K(\omega_{12})$ is

$$K = Av(m_{i1}) \sum_{\dots,m_2\dots} |A_{12}|^2 \Delta \left[\omega_{12} - \sum_{l} \omega_{l}(m_{l2} - m_{l1}) \right].$$
(94)

Here, the dependence of the amplitude A_{12} on all transient (tunneling and above-barrier) states ... m_i ... is determined by Eq. (80) [11, 79], the external summation is performed over all final states ... m_{i2} ... on the surface

$$\sum_{i} \omega_{i}(m_{i2} - m_{i1}) = \omega_{12} \tag{95}$$

in concordance with the law of energy conservation

$$\hbar\Omega = J_1 - J_2 + \hbar\omega_{12} \tag{96}$$

 $(J_1 \text{ and } J_2 \text{ are the electron binding energies in the initial and final states 1 and 2, see also Section 2.4.1 above), <math>Av(m_{i1})$ is the average of the initial states ... m_{i1} ... over the distribution function of Planck. The wavelength λ (see Section 1 above, Figs. 1 and 2) refers to the frequency Ω in Eq. (96) by the formula $\lambda = 2\pi c/\Omega n_{ref}$.

In Eq. (80), let us change the summing over the transient (intermediate) states $...m_{i...}$ in all modes *i* by summing on the surface

$$\sum_{i}\omega_{i}(m_{i}-m_{i1})=\omega_{1}$$
(74a)

(compare with Eq. (74) in Section 2.4.4) and by the sum over all integers ω_1 from $-\infty$ to $+\infty$. Then, taking into account Eq. (89), we obtain the amplitude of electron-charge transfers A_{12} in the following form:

$$A_{12} = c_{12} \sum_{l} \omega_{l} \tilde{p}_{l1} \sum_{\omega_{1} = -\infty}^{\infty} G^{\mathrm{E}}(\omega_{1}, L) P_{l}(m_{i2}, m_{i1}; \omega_{1}),$$
(97)

where the Green's function of the elementary electron transfers $G^{\text{E}}(\omega_1, L)$ is given by Eqs. (91), (92), and (78), and $P_l(m_{l2}, m_{l1}; \omega_1)$ denotes the advanced phonon factor, which, due to the completeness of the system of phonon wave functions $\Phi_{0...m_{l...}}(p)$ (see Eq. (79)), transforms into a Franck-Condon type factor

$$P_{l}^{\text{FC}}(m_{l2}, m_{l1}) = \frac{o(m_{l2}, m_{l1})}{r(m_{l2}, m_{l1})} \prod_{l} r(m_{l2}, m_{l1}) , \qquad (99)$$

if in the Green's function of elementary electron-charge transfers $G^{\rm E}(\omega_1, L)$ entering into Eq. (97), we put the number ω_1 = constant:

$$A_{12} = c_{12} G^{\rm E}(\omega_1 = \text{constant}, L) \sum_l \omega_l \tilde{p}_{l1} P_l^{\rm FC}(m_{l2}, m_{l1}).$$
(100)

The factor $P_l(m_{i2}, m_{i1}; \omega_1)$ (Eq. (98)) take appropriately all possible virtual electron and nuclear movements into account, which create intermediate energy levels $J_1 + \hbar \omega_1$ of the electron.

To calculate the factor $P_l(m_{i2}, m_{i1}; \omega_1)$ and other similar quantities

 $^{^{14}}$ Heat energy $\Delta<0$ matches the acts of luminescence, research of which is beyond the scope of this article. See also the latest reference in this article.

further, we need the Cauchy theorem with respect to the Laurent series:

$$f(w) = \sum_{m=-\infty}^{\infty} b_m (w - w_0)^m, \ b_m = \frac{1}{2\pi \ i} \oint \frac{f(\zeta)}{\left(\zeta - w_0\right)^{m+1}} d\zeta,$$
(101)

where the contour encircles the point w_0 , in the particular case $w_0 = 0$. So, to calculate the factor $P_l(m_{i2}, m_{i1}; \omega_1)$ in Eq. (98), we introduce the

generating polynomials

$$rr(m_{i2}, m_{i1}; u) \equiv r(m_{i2}, m_{i1})r(m_{i1}, m_{i1}) + u^{\omega_i}r(m_{i2}, m_{i1} + 1)r(m_{i1} + 1, m_{i1}) + \dots + u^{-\omega_i}r(m_{i2}, m_{i1} - 1)r(m_{i1} - 1, m_{i1}) + \dots$$
(102)

and

$$ro(m_{l2}, m_{l1}; u) \equiv r(m_{l2}, m_{l1})o(m_{l1}, m_{l1}) + u^{\omega_l}r(m_{l2}, m_{l1} + 1)o(m_{l1} + 1, m_{l1}) + \dots + u^{-\omega_l}r(m_{l2}, m_{l1} - 1)o(m_{l1} - 1, m_{l1}) + \dots,$$

where the Condon integrals r(...) and o(...) are given by Eqs. (82) and (83), correspondingly. Then, using the Cauchy theorem with respect to

the Laurent series (see Eq. (101)), from Eq. (98) we obtain

$$P_{l}(m_{l2}, m_{l1}; \omega_{1}) = \frac{1}{2\pi i} \oint \frac{du}{u^{\omega_{1}+1}} \frac{ro(m_{l2}, m_{l1}; u)}{rr(m_{l2}, m_{l1}; u)} \prod_{l} rr(m_{l2}, m_{l1}; u),$$
(104)

in Eq. (105) will be hereinafter referred to as $R_{l,l'}(\omega_1, \omega'_1; \omega_{12})$. This function describes the electron-phonon dynamics for the individual levels ω_1 and ω'_1 and the phonon modes l and l', which is due to the virtual electron-phonon coupling. Note that the quantity

$$\sum_{l} \omega_{l} \tilde{p}_{l1} \sum_{l'} \omega_{l'} \tilde{p}_{l'1} R_{l,l'} \left(\omega_{1}, \omega_{1}'; \omega_{12} \right) \equiv \phi \left(\omega_{1}, \omega_{1}'; \omega_{12} \right)$$
(108)

(see also the result for the function $\phi(\omega_1, \omega'_1; \omega_{12})$ in Eq. (120) below) describes the electron-phonon dynamics for the individual levels ω_1 and ω'_1 , which is caused by the virtual electron-phonon coupling, taking into account all the phonon modes.

So, using the function $R_{l,l'}(\omega_1, \omega'_1; \omega_{12})$, the formula (105) for the optical absorption *K* is rewritten as follows:

$$K \equiv K(\omega_{12}) = c_{12}^{2}$$

$$\times \sum_{\omega_{1}=-\infty}^{\infty} G^{E}(\omega_{1},L) \sum_{\omega_{1}'=-\infty}^{\infty} G^{E} * (\omega_{1}',L) \sum_{l} \omega_{l} \tilde{p}_{l1} \sum_{l'} \omega_{l'} \tilde{p}_{l'1} R_{l,l'} (\omega_{1},\omega_{1}';\omega_{12})$$
(109)

Substituting the factors $P_l(m_{i2}, m_{i1}; \omega_1)$ and $P_l(m_{i2}, m_{i1}; \omega'_1)$, expressed in terms of contour integrals and polynomials (Eqs. (102) and (103)), from Eqs. (104) and (106) into the expression (107), we have

$$R_{l,l}\left(\omega_{1},\omega_{1}^{'};\omega_{12}\right) = Av(m_{l1})\frac{1}{\left(2\pi i\right)^{2}} \oint \frac{du}{u^{\omega_{1}+1}} \oint \frac{dv}{v^{\omega_{1}^{'}+1}} \sum_{\dots,m_{2}\dots} \Delta \left[\omega_{12} - \sum_{i} \omega_{i}(m_{i2}-m_{i1})\right] \frac{ro(m_{l2},m_{l1};u)ro(m_{l'2},m_{l'1};v)}{rr(m_{l2},m_{l1};u)r(m_{l'2},m_{l'1};v)} \prod_{i} rr(m_{i2},m_{i1};u)rr(m_{l2},m_{i1};v).$$
(110)

(103)

where the contour encircles the point u = 0. Substituting the amplitude A_{12} from Eq. (97) into Eq. (94) for the optical absorption *K*, we have

Further, in Eq. (110) we perform summation over all final states 2, using, as in the derivation of Eq. (104), the Cauchy theorem with respect to the Laurent series (see Eq. (101)). For $l \neq l$ we get

$$R_{l,l}\left(\omega_{1},\omega_{1}^{'};\omega_{12}\right) = Av(m_{l})\frac{1}{(2\pi i)^{3}} \oint \frac{du}{u^{\omega_{1}+1}} \oint \frac{dv}{v^{\omega_{1}^{'}+1}} \oint \frac{dw}{w^{\omega_{12}+1}} rorr(m_{l};u,v,w) rro(m_{l'};u,v,w) \prod_{l} rrrr(m_{l};u,v,w), \tag{111}$$

$$K = c_{12}^{2} \sum_{\omega_{1} = -\infty}^{\infty} G^{E}(\omega_{1}, L) \sum_{\omega_{1}' = -\infty}^{\infty} G^{E} \star (\omega_{1}', L) \sum_{l} \omega_{l} \tilde{p}_{l1} \sum_{l} \omega_{l} \tilde{p}_{l1} Av(m_{i1}) \sum_{\dots, m_{2} \dots} P_{l}(m_{i2}, m_{i1}; \omega_{1}) P_{l}(m_{i2}, m_{i1}; \omega_{1}') \Delta \left[\omega_{12} - \sum_{l} \omega_{l}(m_{i2} - m_{i1}) \right],$$
(105)

where $P_{\ell}(m_{i2}, m_{i1}; \omega'_1)$ is the advanced phonon factor arising from the complex conjugate amplitude A_{12}^* . By the same method that we got the result (104), we get

$$P_{\ell}\left(m_{l2}, m_{l1}; \omega_{1}'\right) = \frac{1}{2\pi i} \oint \frac{dv}{v^{\omega_{1}'+1}} \frac{ro(m_{\ell\,2}, m_{\ell\,1}; v)}{rr(m_{\ell\,2}, m_{\ell\,1}; v)} \prod_{l} rr(m_{l2}, m_{l1}; v).$$
(106)

The quantity

$$Av(m_{i1})\sum_{\dots,m_{2}\dots}P_{i}(m_{i2},m_{i1};\omega_{1})P_{i}(m_{i2},m_{i1};\omega_{1}')\Delta\left[\omega_{12}-\sum_{i}\omega_{i}(m_{i2}-m_{i1})\right]$$
(107)

where the contours encircle the points u = 0, v = 0 and w = 0, correspondingly; the generating polynomials entering here are determined in the following way:

$$rrrr(m_{i1}; u, v, w) \equiv rr(m_{i1}, m_{i1}; u)rr(m_{i1}, m_{i1}; v) + w^{\omega_{i}}rr(m_{i1} + 1, m_{i1}; u)rr(m_{i1} + 1, m_{i1}; v) + ...$$
(112)
$$+ w^{-\omega_{i}}rr(m_{i1} - 1, m_{i1}; u)rr(m_{i1} - 1, m_{i1}; v) + ...,$$

$$rorr(m_{l_{1}};u,v,w)rrrr(m_{l_{1}};u,v,w) \equiv ro(m_{l_{1}},m_{l_{1}};u)rr(m_{l_{1}},m_{l_{1}};v) + w^{\omega_{l}}ro(m_{l_{1}}+1,m_{l_{1}};u)rr(m_{l_{1}}+1,m_{l_{1}};v) + ... + w^{-\omega_{l}}ro(m_{l_{1}}-1,m_{l_{1}};u)rr(m_{l_{1}}-1,m_{l_{1}};v) + ...$$
(113)

Heliyon 5 (2019) e02579

(115)

and

$$rrro(m_{\ell_1}; u, v, w) rrrr(m_{\ell_1}; u, v, w) \equiv rr(m_{\ell_1}, m_{\ell_1}; u) ro(m_{\ell_1}, m_{\ell_1}; v) + w^{\omega_{\ell}} rr(m_{\ell_1} + 1, m_{\ell_1}; u) ro(m_{\ell_1} + 1, m_{\ell_1}; v) + \dots$$
(114)
+ w^{-\omega_{\ell}} rr(m_{\ell_1} - 1, m_{\ell_1}; u) ro(m_{\ell_1} - 1, m_{\ell_1}; v) + \dots

For l = l', the function $R_{ll'}(\omega_1, \omega_1'; \omega_{12})$ is formulated as

where *T* is the absolute temperature. Finally, in the resulting Eq. (117) for the optical absorption *K*, the Green's function of elementary electron transfers $G^{E}(\omega_{1}, L)$ is given by Eqs. (91), (92), and (78), and the constant c_{12} is given by Eq. (90).

2.4.7. The case of non-local phonons For simplicity, we do not take into account local vibrations, and so we

$$R_{l,l}(\omega_1,\omega_1';\omega_{12}) = Av(m_{l1})\frac{1}{(2\pi i)^3} \oint \frac{du}{u^{\omega_1+1}} \oint \frac{dv}{v^{\omega_1'+1}} \oint \frac{dw}{w^{\omega_{12}+1}} roro(m_{l1};u,v,w) \prod_{l} rrrr(m_{l1};u,v,w),$$

where the generating polynomials

$$roro(m_{l1}; u, v, w) rrrr(m_{l1}; u, v, w) \equiv ro(m_{l1}, m_{l1}; u) ro(m_{l1}, m_{l1}; v) + w^{\omega_l} ro(m_{l1} + 1, m_{l1}; u) ro(m_{l1} + 1, m_{l1}; v) + ...$$
(116)
+ $w^{-\omega_l} ro(m_{l1} - 1, m_{l1}; u) ro(m_{l1} - 1, m_{l1}; v) + ...,$

and the generating polynomials $rrrr(m_{i1}; u, v, w)$ are provided by Eq. (112). The total electron-phonon dynamics due to the virtual electron-phonon coupling is taken into account by the sum over all transient (intermediate) levels of the electron energy $\hbar \omega_1$ (and the sum over all phonon modes *l*) in the optical absorption *K*. After the substitution of the functions $R_{l,l}(\omega_1, \omega'_1; \omega_{12})$ and $R_{l,l}(\omega_1, \omega'_1; \omega_{12})$ from Eqs. (111) and (115) into Eq. (109) for *K*, we obtain finally [11, 79].

consider below the average of the initial states $...m_1...$ over the distribution function of Planck,

$$Av(m_1)[P(m_1; u, v, w)R(m_1; u, v, w)]$$
(121)

(see this term in Eq. (117)), merely for non-local vibrations of nuclei (crystal lattice phonons). In this case, we can take advantage of the fact that

$$\tilde{p}_{i} \sim M^{-1/2},$$
 (122)

where *M* is the amount of multipliers in the generating polynomial $P(m_1; u, v, w)R(m_1; u, v, w)$ (see Eqs. (118) and (119)), which is equal to the amount of vibrational degrees of freedom in a condensed medium. Being

$$K = c_{12}^{2} \sum_{\omega_{1} = -\infty}^{\infty} G^{\mathrm{E}}(\omega_{1}, L) \sum_{\omega_{1}^{'} = -\infty}^{\infty} G^{\mathrm{E}} \star \left(\omega_{1}^{'}, L\right) \frac{1}{(2\pi i)^{3}} \oint \frac{du}{u^{\omega_{1}+1}} \oint \frac{dv}{v^{\omega_{1}^{'}+1}} \oint \frac{dw}{w^{\omega_{12}+1}} Av(m_{1}) [P(m_{1}; u, v, w)R(m_{1}; u, v, w)], \tag{117}$$

where the generating polynomial

$$P(m_{1}; u, v, w) \equiv \sum_{l} \sum_{\substack{\ell \neq l}} \omega_{l} \omega_{\ell} \tilde{p}_{\Pi} \tilde{p}_{\ell 1} rorr(m_{l1}; u, v, w) rrro(m_{\ell 1}; u, v, w)$$
$$+ \sum_{l} \omega_{\ell}^{2} \tilde{p}_{\Pi}^{2} roro(m_{l1}; u, v, w)$$
(118)

and the generating polynomial

$$R(m_1; u, v, w) \equiv \prod_{i} rrrr(m_{i1}; u, v, w);$$
(119)

the generating polynomials $rrrr(m_{i1}; u, v, w)$, $rorr(m_{l1}; u, v, w)$, $rrro(m_{l1}; u, v, w)$ and $roro(m_{l1}; u, v, w)$ are provided by Eqs. (112), (113), (114), and (116), respectively (see also the polynomials in Eqs. (102) and (103) above). Note that the result for the function $\phi(\omega_1, \omega'_1; \omega_{12})$, which describes the electron-phonon dynamics for the individual levels ω_1 and ω'_1 because of the virtual electron-phonon coupling (with allowance for all phonon modes) and being determined above according to Eq. (108), is the following:

$$\phi(\omega_{1},\omega_{1}';\omega_{12}) \equiv \phi(\omega_{1},\omega_{1}';T;\omega_{12}) \\
= \frac{1}{(2\pi i)^{3}} \oint \frac{du}{u^{\omega_{1}+1}} \oint \frac{dv}{v^{\omega_{1}'+1}} \oint \frac{dw}{w^{\omega_{12}+1}} Av(m_{1}) \\
\times [P(m_{1};u,v,w)R(m_{1};u,v,w)],$$
(120)

mindful of the limiting transition $M \to \infty$ in the final result, it suffices to hold terms in the polynomial $P(m_1; u, v, w)R(m_1; u, v, w)$ pending order M^{-1} inclusively. The Condon integrals in Eqs. (82) and (83) are easily computed by expanding in powers of \tilde{p}_i , and we obtain

$$r(m_{i1}, m_{i1}) = 1 - \frac{\tilde{p}_i^2}{2} \left(m_{i1} + \frac{1}{2} \right) + \dots, \quad r(m_{i1} + 1, m_{i1}) = \tilde{p}_i \sqrt{\frac{m_{i1} + 1}{2}} + \dots,$$

$$r(m_{i1} - 1, m_{i1}) = -\tilde{p}_i \sqrt{\frac{m_{i1}}{2}} + \dots; \qquad (123)$$

$$o(m_{l1}, m_{l1}) = \frac{\tilde{p}_l}{2} + \dots, \quad o(m_{l1} + 1, m_{l1}) = \sqrt{\frac{m_{l1} + 1}{2}} + \dots,$$
$$o(m_{l1} - 1, m_{l1}) = \sqrt{\frac{m_{l1}}{2}} + \dots.$$
(124)

Considering Eqs. (122), (123), and (124), it is readily seen that transitions at which one of the phonon occupation numbers changes by over 1 do not contribute to the polynomial $P(m_1; u, v, w)R(m_1; u, v, w)$. Substituting the results (123) and (124) into the polynomials (102) and (103) and then substituting the polynomials (102) and (103) into the polynomials (112, 113, 114) and (116), we have the following results for generating polynomials of three variables:

$$rrrr(m_{i1}; u, v, w) = 1 - (\tilde{p}_{i1}^2 + \tilde{p}_{i2}^2) \left(m_{i1} + \frac{1}{2} \right) + \frac{1}{2} \tilde{p}_{i1}^2 [(m_{i1} + 1)(uvw)^{\omega_i} + m_{i1}(uvw)^{-\omega_i}] + \frac{1}{2} \tilde{p}_{i2}^2 [(m_{i1} + 1)w^{\omega_i} + m_{i1}w^{-\omega_i}] - \frac{1}{2} \tilde{p}_{i1} \tilde{p}_{i2} [(m_{i1} + 1)(u^{\omega_i} + v^{\omega_i})(w^{\omega_i} - 1) + m_{i1}(u^{-\omega_i} + v^{-\omega_i})(w^{-\omega_i} - 1)],$$

$$(125)$$

$$rorr(m_{l1}; u, v, w) rrrr(m_{l1}; u, v, w) = \frac{p_{l1}}{2} [1 + (m_{l1} + 1)(uvw)^{\omega_l} - m_{l1}(uvw)^{-\omega_l}] - \frac{\tilde{p}_{l2}}{2} [(m_{l1} + 1)u^{\omega_l}(w^{\omega_l} - 1) - m_{l1}u^{-\omega_l}(w^{-\omega_l} - 1)],$$
(126)

$$rrro(m_{l1}; u, v, w) rrrr(m_{l1}; u, v, w) = \frac{\tilde{p}_{l1}}{2} [1 + (m_{l1} + 1)(vuw)^{\omega_l} - m_{l1}(vuw)^{-\omega_l}] - \frac{\tilde{p}_{l2}}{2} [(m_{l1} + 1)v^{\omega_l}(w^{\omega_l} - 1) - m_{l1}v^{-\omega_l}(w^{-\omega_l} - 1)] = rorr(m_{l1}; v, u, w) rrrr(m_{l1}; v, u, w) = rorr(m_{l1}; v, u, w) rrrr(m_{l1}; u, v, w),$$
(127)

 $roro(m_{l1}; v, u, w)rrrr(m_{l1}; u, v, w) = \frac{1}{2} [(m_{l1} + 1)(uvw)^{\omega_l} + m_{l1}(uvw)^{-\omega_l}].$ (128)

By virtue of the relation (122) and in accordance with Eqs. (125), (126), (127), and (128), the generating polynomial $P(m_1; u, v, w)R(m_1; u, v, w)$ (Eqs. (117, 118, 119)) can be formulated as the product of $M \to \infty$ statistically independent multipliers that are linearly dependent on m_1 . Consequently, the operation of averaging $P(m_1; u, v, w)R(m_1; u, v, w)$ is reduced to replacement of all m_{r1} and m_{l1} by their average equilibrium values $\overline{m}_{r1,l1}$ (Planck's distribution function):

$$\overline{m}_{l,l,l} = \left[\exp(\hbar\omega_{l,l} / k_{\rm B}T) - 1 \right]^{-1}.$$
(129)

As a result, the generating functions $R(\overline{m}_1; u, v, w)$ and $P(\overline{m}_1; u, v, w)$ and optical absorption K [11, 79] are expressed as follows:

$$P(\overline{m}_{1}; u, v, w) = \frac{1}{4} \left\{ \sum_{l} \omega_{l} \tilde{p}_{l}^{2} [1 + (\overline{m}_{l1} + 1)(uvw)^{\omega_{l}} - \overline{m}_{l1}(uvw)^{-\omega_{l}}] \right\}^{2} + \frac{1}{2} \sum_{l} \omega_{l}^{2} \tilde{p}_{l}^{2} [(\overline{m}_{l1} + 1)(uvw)^{\omega_{l}} + \overline{m}_{l1}(uvw)^{-\omega_{l}}].$$
(132)

We note that for simplicity, the small terms $\tilde{p}_{i1}\tilde{p}_{i2}\propto_L^1$ and $\tilde{p}_{l1}\tilde{p}_{l2}\propto_L^1$, related to the phonon correlations, are neglected in the generating functions $R(\overline{m}_1; u, v, w)$ and $P(\overline{m}_1; u, v, w)$.

2.4.8. The analytical result for the shape of the optical absorption band

In the expression (130, 131, 132) for the optical absorption *K*, we expand the logarithm of the Green's function of electron transfers $G^{\rm E}(\omega_1, L) \equiv G^{\rm E}(J_1 + \hbar\omega_1; L)$ (see Eqs. (91) and (92) in Section 2.4.5) with respect to the intermediate levels of the electron energy $\hbar\omega_1$ in the vicinity of the donor level J_1 . Restricting ourselves to the linear approximation, we have

$$G^{\mathrm{E}}(J_1 + \hbar\omega_1; L) \approx G^{\mathrm{E}}(J_1; L) \exp(-t\omega_1), \tag{133}$$

$$t = -\left[\partial \ln G^{\rm E}(J_1 + \hbar \omega_1; L) / \partial \omega_1\right]_{\omega_1 = 0}.$$
(134)

Substituting the approximate expression (133) into the exact result (130, 131, 132) and applying the Cauchy theorem with respect to the Laurent series (see Eq. (101)), we get [11, 79].

$$K = c_{12}^{2} |G^{\rm E}(J_{1};L)|^{2} \frac{1}{2\pi i} \oint \frac{dw}{w^{\omega_{12}+1}} P[\overline{m}_{1}; u = \exp(-t), v$$

= $\exp(-t), w] R[\overline{m}_{1}; u = \exp(-t), v = \exp(-t), w].$ (135)

It goes without saying that the application of the linear approximation to $\ln[G^{E}(J_{1} + \hbar\omega_{1}; L)]$ (Eq. (133)) imposes restrictions on the problem parameters; however, it can be shown that these restrictions are not severe [52].

Methods for calculating the contour integral in Eq. (135) are similar to those applied in the standard theory of many-phonon transitions (Ref. [72]). In Eq. (135), the generating function $R[\overline{m}_1; u = \exp(-t), v = \exp(-t), w]$ (Eq. (131)) is exponential with

$$K = c_{12}^{2} \sum_{\omega_{1}=-\infty}^{\infty} \sum_{\omega_{1}'=-\infty}^{\infty} G^{\mathrm{E}}(\omega_{1},L) G^{\mathrm{E}} \star (\omega_{1}',L) \frac{1}{(2\pi i)^{3}} \oint \frac{du}{u^{\omega_{1}+1}} \oint \frac{dv}{v^{\omega_{1}'+1}} \oint \frac{dw}{w^{\omega_{12}+1}} P(\overline{m}_{1};u,v,w) R(\overline{m}_{1};u,v,w),$$
(130)

$$R(\overline{m}_{1}; u, v, w) = \exp\left\{-\sum_{i} \tilde{p}_{i}^{2}(2\overline{m}_{i1}+1) + \frac{1}{2}\sum_{i} \tilde{p}_{i}^{2}[(\overline{m}_{i1}+1)(u^{\omega_{i}}v^{\omega_{i}}+1)w^{\omega_{i}} + \overline{m}_{i1}(u^{-\omega_{i}}v^{-\omega_{i}}+1)w^{-\omega_{i}}]\right\},$$
(131)

respect to the variable *w*, while the generating function $P[\overline{m}_1; u = \exp(-t), v = \exp(-t), w]$ (Eq. (132)) is a power function in *w*. Consequently, in a greater in modulus thermal effect $\hbar \omega_{12}$, when the phonon frequencies

$$\omega_{l,l} < < |\omega_{12}|, \tag{136}$$

we can put ω_l equal to zero in quantity $w^{\pm \omega_l}$, which appears in $P[\overline{m}_1; u = \exp(-t), v = \exp(-t), w]$.¹⁵ Thus, *P* is not now dependent on *w*, and in Eq. (135) it can be brought outside the integral character in the point w = 1 [11, 79]:

$$K = c_{12}^2 \left| G^{\rm E}(J_1;L) \right|^2 P[\overline{m}_1; u = \exp(-t), v = \exp(-t), w = 1] \frac{1}{2\pi i} \oint \frac{dw}{w^{\omega_{12}+1}} R[\overline{m}_1; u = \exp(-t), v = \exp(-t), w].$$
(137)

¹⁵ Inter alia, the restriction (136) here means that we neglect the well-known zero-phonon line (on zero-phonon lines, see Refs. [65, 66, 84]).

V.V. Egorov

Within the framework of Einstein's model of the vibrations of nuclei $\omega_t = \text{constant} \equiv \omega$, in Eq. (137) the contour integral is calculated exactly. In order to obtain the result, one can use, for example, the well-known expansion in a series [85]:

$$\exp\left(Az + \frac{B}{z}\right) = \sum_{n=-\infty}^{\infty} z^n \left(\frac{A}{B}\right)^{\frac{n}{2}} I_n\left(2\sqrt{AB}\right),\tag{138}$$

where $I_n(...)$ is the modified Bessel function. Applying once again the Cauchy theorem with respect to the Laurent series (see Eq. (101)), from Eq. (137), taking Eq. (138) into account, we get

$$K = c_{12}^2 |G^{\rm E}(J_1;L)|^2 P[\overline{m}_1; u = \exp(-t), v = \exp(-t),$$

$$w = 1] \exp\left[-\frac{2E}{\hbar\omega} \operatorname{cth} \beta_T + (\beta_T - t)\frac{\omega_{12}}{\omega}\right] I_{\frac{\omega_{12}}{\omega}} \left(\frac{2E \operatorname{ch} t}{\hbar\omega \operatorname{sh} \beta_T}\right),$$
(139)

where $\beta_T \equiv \hbar \omega / 2k_{\rm B}T$. Further, using the well-known asymptotic formula [85, 86]

$$I_{\kappa}(u) \approx \frac{1}{\sqrt{2\pi u}} \exp\left(u - \frac{\kappa^2}{2u}\right), \ u > 1, \ \kappa \le u,$$
(140)

as a result, we have optical absorption *K*, which is entirely expressed in elementary functions [10, 11, 12, 79]:

$$K = K_0 \exp(Y), \tag{141}$$

$$Y = \frac{1}{2} \ln \left(\frac{\omega \tau \sinh \beta_T}{4\pi \cosh t} \right) - \frac{2}{\omega \tau} \left(\coth \beta_T - \frac{\cosh t}{\sinh \beta_T} \right) + (\beta_T - t) \frac{1}{\omega \tau \Theta} - \frac{\sinh \beta_T}{4\omega \tau \Theta^2 \cosh t},$$
(142)

$$1 < < \frac{1}{\omega\tau \ \Theta} \le \frac{2\cosh t}{\omega\tau \ \sinh \beta_{T}},\tag{143}$$

where

$$t = \frac{\omega \tau_{\rm e}}{\theta} \left[\frac{AC + BD}{A^2 + B^2} + \frac{2\Theta(\Theta - 1)}{(\Theta - 1)^2 + (\Theta/\theta_0)^2} + \frac{\theta_0^2}{\theta_0^2 + 1} \right],$$
(144)

$$|\theta_0| > > \frac{E}{2J_1},\tag{145}$$

$$\theta \equiv \frac{\tau_{\rm e}}{\tau} = \frac{L E}{\hbar \sqrt{2J_1/m}}, \ \Theta \equiv \frac{\tau'}{\tau} = \frac{E}{\Delta}, \ \theta_0 \equiv \frac{\tau_0}{\tau} = \frac{E}{\gamma},$$
(146)

$$\tau_{\rm e} = \frac{L}{\sqrt{2J_1/m}}, \ \tau = \frac{\hbar}{E}, \ \tau' = \frac{\hbar}{\Delta}, \ \tau_0 = \frac{\hbar}{\gamma}.$$
(147)

Here, we use the notation

$$A = \cos\left(\frac{\theta}{\theta_0}\right) + \Lambda + \left(\frac{1}{\theta_0}\right)^2 \nu, \ B = \sin\left(\frac{\theta}{\theta_0}\right) + \frac{1}{\theta_0}\mu, \tag{148}$$

$$C = \theta \left[\cos\left(\frac{\theta}{\theta_0}\right) - \frac{1 - \xi^2}{2\theta_0} \sin\left(\frac{\theta}{\theta_0}\right) \right] + \mu,$$
(149)

$$D = \theta \left[\sin\left(\frac{\theta}{\theta_0}\right) + \frac{1 - \xi^2}{2\theta_0} \cos\left(\frac{\theta}{\theta_0}\right) \right] - \frac{2}{\theta_0} \nu,$$
(150)

$$\xi \equiv \left(1 - \frac{E}{J_1}\right)^{1/2} (J_1 > E \text{ by definition}), \tag{151}$$

and where we finally have



Fig. 3. The dynamics of quantum-classical transitions in the neighborhood of the singular point $(\overline{\Omega} = 1, \ \theta_0 = \infty)$ $[(\Delta = E, \ \gamma = 0)]$ is demonstrated by the behaviour of the quantity $t = t(\overline{\Omega}, \theta_0)$ (see Eq. (144)) [11]. For simplification, it is formally assumed here that $J_2 - J_1 = 0$. Then $\Theta^{-1} = \hbar \Omega / E \equiv \overline{\Omega}$. The following system parameters are used: $J_1 = 5$ eV, E = 1 eV, $\mu = m_e, \ \omega = 5 \times 10^{13} \text{ s}^{-1}$ and $L = L^* \approx 0.44$ nm (transferon resonance, Eq. (161)).

$$\Lambda = -\left(\boldsymbol{\Theta} - 1\right)^{2} \varepsilon + \left[\frac{(\boldsymbol{\Theta} - 1)\boldsymbol{\theta}}{\rho} + \boldsymbol{\Theta}(\boldsymbol{\Theta} - 2)\right] \varepsilon^{\frac{1-\rho}{1-\varepsilon}},$$
(152)

$$\mu = 2\Theta(\Theta - 1)\varepsilon - \left[\frac{(2\Theta - 1)\theta}{\rho} + 2\Theta(\Theta - 1)\right]\varepsilon^{\frac{1-\rho}{1-\varepsilon}},$$
(153)

$$\nu = \Theta \left[\Theta \varepsilon - \left(\frac{\theta}{\rho} + \Theta \right) \varepsilon^{\frac{1-\rho}{1-\varepsilon}} \right], \tag{154}$$

$$r \equiv \exp\left(\frac{2\theta}{1+\xi}\right), \ \rho \equiv \sqrt{\xi^2 + \frac{1-\xi^2}{\Theta}}.$$
 (155)

The factor K_0 becomes

$$K_0 = K_0^e K_0^p, (156)$$

where

۶

$$K_{0}^{e} = \frac{2\tau^{3}J_{1}}{m} \frac{(A^{2} + B^{2})\rho^{3}\Theta^{4}\xi}{\theta^{2} \left[(\Theta - 1)^{2} + \left(\frac{\Theta}{\theta_{0}}\right)^{2} \right]^{2} \left[1 + \left(\frac{1}{\theta_{0}}\right)^{2} \right]} \exp\left(-\frac{4\theta}{1 - \xi^{2}}\right)$$
(157)

and

$$K_0^p = \frac{1}{\omega\tau} \left[1 + \frac{\sinh(\beta_T - 2t)}{\sinh\beta_T} \right]^2 + \frac{\cosh(\beta_T - 2t)}{\sinh\beta_T}.$$
 (158)

Conditions (143) and (145) are not any significant restrictions on the characteristics of the system and associated with points of regular approximations done in the computations [11, 79]. Inequalities (143) follow from conditions in formula (140). Inequality (145) follows from the expansion of the radical in a series by smallness of $|\gamma|/J_1$ in the Gamow exponent (see Section 2.4.4, formula (78) for $\omega_1 = 0$), which is limited to the zero-order approximation. The scaling times offered by Eq. (147) regulate the chaotic dynamics of quantum-classical transitions. We consider them at length in another place [12, 13, 14, 28, 79]. Here, they are discussed in a few words.

A crucial component in the dynamics of quantum-classical transitions is a numerical correlation between the time

$$\tau_{\rm e} = \frac{L}{\sqrt{2J_1/m}} \tag{159}$$

and the time

$$\tau = \frac{\hbar}{E},\tag{160}$$

that are elements in Eqs. (141), (142), (143), (144), (145), (146), (147), (148), (149), (150), (151), (152), (153), (154), (155), (156), (157), and (158) for computing optical absorption spectra (Eq. (147)). The time τ_e is the specific time of an electron movement between the donor and the acceptor, set apart by distance *L*. The time τ is the specific time of the environmental nuclear reorganization. In the case

 $(2\tau_e)^{-1} = \tau^{-1} \tag{161}$

the so-called transferon resonance [12, 13] occurs between the respective frequencies $(2\tau_e)^{-1}$ and τ^{-1} . We considered one of the implications of the resonance under discussion as an example in Ref. [18], Section 2.4 (optical absorption spectra as solution-dependent [17, 44]). Other its impacts are considered in Refs. [10, 11, 12, 14, 17, 28]. Farther, in Eq. (147), the time $\tau' = \frac{\hbar}{4}$ is the specific time of transformation of the energy $\hbar \Omega$ of light into the excitation energy $J_1 - J_2$ of electron and the heat energy $\Delta \equiv \hbar \omega_{12}$ in elementary electron-transfer processes (see Eq. (96)), and the time $\tau_0 =$ $\frac{\hbar}{r}$ is the specific time of transformation of a movement (energy) of electron into a movement (energy) of nuclear reorganization ($\gamma > 0$) and/or of the reverse acts ($\gamma < 0$) in the transient dozy-chaos state (the sign of γ is discussed below). The non-dimensional parameters θ , θ , θ_0 (Eq. (146)) are related to the aforementioned specific times τ_{e} , τ' , τ_{0} , which are divided by the specific time τ (Eq. (160)). The quantities $\Lambda = \Lambda(\theta, \Theta), \mu =$ $\mu(\theta, \Theta)$ and $\nu = \nu(\theta, \Theta)$ (see Eqs. (152), (153), and (154)), being independent of the parameter θ_0 and the dozy-chaos energy γ , do not involve the chaos of the electron-nuclear movements in the transient state and describe merely the regular motion of the electron and nuclei. The functions $A = A(\theta, \Theta, \theta_0), B = B(\theta, \Theta, \theta_0), C = C(\theta, \Theta, \theta_0)$ and $D = D(\theta, \Theta, \theta_0)$ (see Eqs. (148), (149), and (150)), being dependent on the parameter θ_0 , already involve the chaos of the electron-nuclear movements in the transient state. The Eqs. (144), (145), (146), (147), (148), (149), (150), (151), (152), (153), (154), (155), and (157) are independent of the parameter $\beta_T \equiv \hbar \omega / 2k_{\rm B}T$ and the absolute temperature *T*, therefore, they do not include the averaging over the equilibrium distribution of the initial states in the ensemble of the donor-acceptor systems. By contrast, the Eqs. (142) and (158), being dependent on the parameter $\beta_T \equiv$ $\hbar\omega/2k_{\rm B}T$, already include the averaging over the equilibrium distribution of the initial states. All donor-acceptor systems in the ensemble have their own values of the parameter Θ , the corresponding thermal effect Δ and absorption frequency Ω of light (see Eq. (96)). These values, together with the other parameters of a donor-acceptor system in the ensemble, determine the position, width, intensity and shape of optical absorption bands. From the formula $\tau' = \frac{\hbar}{\Lambda}$ for the specific time of transformation of the light energy $\hbar\Omega$ (see Eq. (147)) and from the law of energy conservation (see Eq. (96)) it follows that the dynamics of producing the shape of optical bands is fastest in the high-frequency tail of the optical bands and is slowest in their low-frequency tail. Lastly, in Eq. (157), $\exp[-4\theta/(1-\xi^2)] \equiv \exp(-2L/a)$ is the Gamow tunnel factor ($a \equiv$ $\hbar/\sqrt{2mJ_1}$).

The quantity $K = K(\Theta, \theta_0)$ (Eq. (141)) and the corresponding optical extinction (Eq. (93)) have a singularity at the point ($\Delta = E$, $\gamma = 0$) or ($\Theta = 1$, $\theta_0 = \infty$). The nature of this singularity is controlled by singularities of the functions $t = t(\Theta, \theta_0)$ in Eq. (144) and $K_0^e = K_0^e(\Theta, \theta_0)$ in Eq. (157). The singularity in the function $K_0^e = K_0^e(\Theta, \theta_0)$ is removable:

$$\frac{K_0^e(\theta=1,\ \theta_0\to\infty)}{2\tau^3 J_1/m} = \frac{\xi}{\theta^2} \left[\exp\left(\frac{2\theta}{1+\xi}\right) - \frac{\theta^2}{2} - \theta - 1 \right]^2 \exp\left(-\frac{4\theta}{1-\xi^2}\right).$$
(162)

In the function $t = t(\Theta, \theta_0)$, the singularity at the point $(\Theta = 1, \theta_0 = \infty)$ is irremovable. The behavior of the function $t = t(\Theta, \theta_0)$ in the neighborhood of $(\overline{\Omega} \equiv \Theta^{-1} = 1, \theta_0 = \infty)$ is shown in Fig. 3. Must be noted that the result (141, 142, 143, 144, 145, 146, 147, 148, 149, 150, 151, 152, 153, 154, 155, 156, 157, 158) is invariable if you change the sign of γ . The invariancy is consistent with the physical case that both the virtual acts of transformation of electron movement (energy) into nuclear reorganization movement (energy) and the reverse acts occure in the transient dozy-chaos state [14, 15, 18, 28, 79]. For definiteness, we set $\gamma > 0$ in Fig. 3 and hereinafter.

2.4.9. Limit to standard result

The limiting transition from the optical absorption K given by Eqs. (141), (142), (143), (144), (145), (146), (147), (148), (149), (150), (151), (152), (153), (154), (155), (156), (157), (158), and (96) to the corresponding result given by the standard many-phonon theory [72] can be a priory implemented by tending the dozy-chaos energy γ to either zero or infinity, ¹⁶ but the quantity *K* turns out to be infinity for $\gamma \rightarrow 0$ and zero for $\gamma \to \infty$. The physical meaning of $K(\gamma \to 0) \to \infty$ beyond the adiabatic approximation is interpreted as a consequence of the incomparability of masses of the electron and its environmental nuclei (see Section 2.4.3). The physical meaning of $K(\gamma \rightarrow \infty) \rightarrow 0$ (see Fig. 4 in Section 2.6 below) is determined by the failure of electron transitions coupled to nuclear reorganizations when in the transient state the motion of nuclei is absolutely chaotic (random), hat is, when in the electron-nuclear system the internal friction is infinite. Given this, we can remove γ from Eqs. (141), (142), (143), (144), (145), (146), (147), (148), (149), (150), (151), (152), (153), (154), (155), (156), (157), (158), and (96) and get the standard result by tending γ to infinity in the equation for t (t \rightarrow 0; Fig. 3, where $\theta_0 = E/\gamma$ under Eq. (146)) and to zero in K_0^e (Eq. (162)). The standard type equation for K ($k_{\rm B}T > \hbar\omega/2$) is thus obtained [11]:

$$K = \frac{a^2 \hbar}{\sqrt{4\pi \lambda_r k_B T}} \exp\left(-\frac{2L}{a}\right) \exp\left[-\frac{(\Delta - \lambda_r)^2}{4\lambda_r k_B T}\right],$$
(163)

where $\lambda_r \equiv 2E$. A formula of this type was obtained by Marcus in his electron-transfer model [87, 88, 89, 90, 91, 92] and is often called the Marcus formula, and the energy λ_r is called the reorganization energy of Marcus. Similar and more general formulas were previously obtained in the theory of many-phonon transitions (see Refs. [65, 72]) for optical transitions by Huang and Rhys [93] and Pekar [73, 74, 86] (see also Lax [94] and Krivoglaz and Pekar [76]) and for nonradiative transitions by Huang and Rhys [93] and Krivoglaz [77].

2.4.10. Quantum-classical mechanics as applied to both solids and liquids

Although the basic principles of the quantum-classical mechanics of elementary electron transfers were given in this article in terms corresponding to solids, nowhere was the regular periodic configuration of

¹⁶ Condition (145) does not allow us to make this limit in our result (141, 142, 143, 144, 145, 146, 147, 148, 149, 150, 151, 152, 153, 154, 155, 156, 157, 158) as the gamma tends to infinity $(\gamma \to \infty)$. However, this condition is not fundamental and is related only to routine simplification of calculations, namely it is connected, as previously indicated, with the expansion of the radical into a series of smallness $|\gamma|/J_1$ in the exponent of the Gamow exponent, which is limited to the zero-order approximation. If this simplification is refused, then the result of the limiting transition under discussion is the same as here: $K(\gamma \to \infty) \to 0$. In this connection, the rejection of the aforementioned expansion only in an uninteresting, from the perspective of applications, asymptotic parameter range (that is, in the region $K(\gamma \to \infty) \to 0$).



metion plays the role of dozy endos

Fig. 4. Singularity in the rate of molecular "quantum" transitions: a potential box with a moveable wall (*a*) and the optical absorption band shape dependant on the dozy chaos accessible to a certain "quantum" transition (*b*); the pronounced peak (J-band) matches the least dozy chaos [15, 18]. (Original citation) — Reproduced by permission of The Royal Society of Chemistry.

atoms (nuclei) assumed. It was assumed only that the atoms make harmonic oscillations around some still equilibrium states. But such a condition also occurs in liquids for quite long periods of time until shaking happens due to a strong statistical fluctuation of the oscillations and new equilibrium states of the atoms are set up. Since the shaking time $\tau_T \cong \hbar/k_{\rm B}T$ greatly exceeds the elementary electron transfer time $\tau \cong \tau = \hbar/E$ (Section 2.4.8, Eq. (160)), because of the irresistible condition $k_{\rm B}T < < E$ (at room temperature $k_{\rm B}T = 1/40$ eV and $E = (1 \div 0.1)$ eV), our theory, such as the standard theory of many-phonon transitions [72, 73, 74], can be applied to both solids and liquids [11]. A clear confirmation of this is, for example, our successful theoretical explanations for the optical spectra in polymethine dyes and their aggregates in liquid solutions (Section 1).

2.5. Internal inconsistency of the Franck-Condon principle

As mentioned in Section 2.1, just as in quantum-classical mechanics, the transient state in the standard Franck-Condon picture is essentially classical: the classical motion of the nuclei is carried out to the turning point, or in other words, "towards the quantum transition." In the standard theory, this qualitative Franck-Condon picture of the dynamics of the transient state does not have a formalized description. When we attempt to formalize this picture within the framework of quantum mechanics, we immediately obtain a singularity in the rates of molecular quantum transitions. This is because we use only quantum mechanics to describe molecular "quantum" transitions, whereas in the qualitative Franck-Condon picture, in addition to the quantum transition, we also have the classical motion of the nuclei "towards it." Therefore, to eliminate the singularity, a certain paradigm of classicality must be introduced, in one way or another, into quantum mechanics. The simplest way to do this in the formal apparatus of quantum mechanics is to replace the infinitesimal imaginary additive in the energy denominator of the spectral representation of the total Green's function of the system by its finite value and thus come to quantum-classical mechanics. In the aforementioned reasoning, we did not have to resort, as we did earlier in the text of the article, to the argument that the mass of the nuclei is incomparably larger than the mass of the electrons (in this regard, see footnote 10 in Section 2.4.3). It was sufficient to refer to the internal inconsistency of the Franck-Condon picture, which is related with the fact that in theory, the qualitative Franck-Condon picture of the dynamics of the transient state does not have an adequate mathematical formulation.¹⁷

We previously discussed the internal inconsistency of the Franck-Condon principle related to the inability to formalize, within the framework of quantum mechanics, the qualitative quantum-classical picture of optical transitions in molecules, which the Franck-Condon principle offers. A similar internal inconsistency is inherent in the Franck-Condon principle when it is used in the standard theory of nonradiative elementary electron transfers in condensed media, namely, in Marcus' theory. In the qualitative picture of Marcus, it is assumed that the electron energy on the donor and on the acceptor in each individual donor-acceptor pair is equalized due to fluctuations in the classical motion of the nuclear subsystem (due to thermal fluctuations of the nuclear coordinates) before the quantum transition (transfer) of an electron occurs from the donor to the acceptor [91, 92]. In other words, the quantum transition occurs at the "point" of intersection of the electronic terms of reagents and products,¹⁸ which is achieved as a result of the classical reorganization of the nuclear subsystem, and each individual donor-acceptor pair immersed in the medium acts, as in the case of transitions in individual molecules, as a simple quantum-classical system.

¹⁷ The incomparability of the masses of electrons and nuclei is present in the Franck-Condon picture and "substantiates" it, but here, first of all, we start in our reasoning from the internal inconsistency of the Franck-Condon picture itself, when the incomparability of the masses of electrons and nuclei recedes into the background.

 $^{^{18}}$ For a system with *M* nuclear coordinates, this intersection occurs on an (*M* – 1)-dimensional surface.

An attempt to formalize the qualitative picture of Marcus in the framework of quantum mechanics, which we essentially undertook at the initial stage of building quantum-classical (dozy-chaos) mechanics, leads to a singularity in the rates of transitions (transfers). Damping this singularity, that is, introducing chaos into the transient state of each individual donor-acceptor pair, we obtain the quantum-classical (dozy-chaos) mechanics of elementary electron transfers in condensed matter, which is formulated in Section 2.4. The result of the rate constant in Marcus' theory is obtained from the result of the rate constant in dozy-chaos mechanics when we turn to strong chaos [11] ($\gamma \ge E$; see also the limiting transition in Section 2.4.9). It follows that in the problem of elementary electron transfers, the Franck-Condon principle acts as an effective simulator of strong dozy chaos, just as in the problem of "quantum" transitions in molecules, and the Marcus model acts as an effective simulator of quantum-classical mechanics of electron transfers in strong dozy chaos.

2.6. Explanation for "pedestrians" why quantum mechanics ceases to work in molecular and chemical physics

Quantum mechanics was invented in the last century to explain the quantum jumps of an electron in an atom. Let's assume that the atom can be schematically modeled by a one-dimensional potential box (Fig. 4a). Then, what is the model of a molecule in this scheme [49]? In a molecule in the process of a quantum transition from one state to another state, the nuclear subsystem is reorganized, that is, the equilibrium positions of the nuclear vibrations are adjusted to the new distribution of the electron charge in the final state. What does this adjustment or reorganization correspond to in the scheme of the one-dimensional potential box [49]? Obviously, this reorganization corresponds to the motion of at least one of the two walls of the potential box during the quantum transition. There are only two options of the motion of this wall chosen by us. The first option: the wall moves freely, that is, without friction. The second option: the wall moves with friction [14, 15, 18, 28, 49].

Initially, consider the first option. In the course of the electron transition, for example, from the ground state, where the probability of finding the electron is greatest in the middle of the box, to the first excited state, where this probability becomes larger both near the stationary and movable walls, the electron pressure on the walls arises. This pressure sets the free wall in motion. Due to the very rapid motion of the free wall, the width of the potential box increases very rapidly and unlimitedly. As a result, the energy level of the excited state falls very rapidly straight down to coincidence with the energy level of the ground state; this obviously leads to a singularity (to infinity) in the quantum transition rate [14, 15, 18, 28, 49]. From this schematic analysis it follows that, within the framework of quantum mechanics, the full-fledged joint motion of an electron and nuclei in a molecule can not be regular, but it can only be singular. (In the adiabatic approximation [53] (see Section 2.3) in molecular physics and quantum chemistry, the motion of electrons is not dynamically full-fledged: they only form an electric potential in which the nuclei make their oscillations.)

Let us now turn to the second option, in which the wall of the potential box moves with friction. In this case, in the process of a quantum transition, because of the presence of friction, the wall will not have time to go to infinity, and therefore the rate of the electron transition becomes a normal finite value. In a real molecule, this friction corresponds to chaos, which provokes a very light electron in the oscillatory movements of massive nuclei "with the aim" to control their movements during molecular quantum transitions. As previously mentioned (Sections 1, 2.1, and 2.4.3), this chaos is called dozy chaos [14, 28], since it arises merely in a transient state and is absent in the initial and final states. Because of chaos (dozy chaos) in the transient dynamic state, a continuous spectrum of energy in this state appears (see also Sections 1 and 2.1), which is a sign of classical mechanics [16, 18]. In other words, the whole theory of molecular quantum transitions ceased to be quantum mechanics.

by a potential box with a moveable wall (Fig. 4a) [14, 15, 18, 28, 49]. The wall is attached to the abscissa axis by an easily moveable hinge and can shift along the axis with some friction or without friction. Suchlike a wall imitates the nuclear reorganization of the environment in the elementary electron transfers, where dozy chaos acts as friction. In the theory [10, 11, 12, 13, 14, 15, 28, 79], this results in the displacement of the dozy-chaos dependent optical absorption band to the red spectral region and its narrowing (Fig. 4b). The width and the intensity of the optical band (see Fig. 4) are controlled by the proportion $\theta_0 = E/\gamma$ (see Eq. (146)), where *E* is the reorganization energy and γ is the dozy-chaos energy. The smaller the magnitude of γ is, the sharper "spike" in the dynamic function $t = t(\overline{\Omega}, \theta_0)$ (see Fig. 3, $\gamma \ll E$ or $\theta_0 \gg 1$), the higher the level of organization of the elementary electron transfer, and the smaller the width and larger the intensity of the optical band (see Fig. 4). The red maximum drift in Fig. 4b can be directly understood by considering the potential box with a moveable wall (Fig. 4a). This drift can be also realized from the band shape behavior with the change (decrease) in the nuclear reorganization energy in the standard theory, if the reorganization energy is thought of as a complex quantity, and the dozy-chaos energy γ acts as its imaginary part (see details in Ref. [15]).

constants of electron-nuclear(-oscillatory) transitions, exemplified here

by elementary electron transfers in condensed matter, can be elucidated

Heliyon 5 (2019) e02579

Consider the basics of quantum mechanics from another point of view. Molecules and atoms are similar to each other. Their similarity lies in the fact that the structural elements of both microsystems are nuclei and electrons. The main difference between these microsystems is the number of nuclei: atoms are electron-nuclear microsystems that have only one single nucleus, while molecules are electron-nuclear microsystems that generally have a large number of nuclei (at least two). It is this well-known and trivial fact that ultimately serves as the divide between quantum and quantum-classical mechanics. Quantum mechanics was created at one time to describe the electronic structure of atoms and quantum transitions of electrons between different states of this structure. In stationary states of atoms, due to the massive and single nucleus, the electronic structure is formed as a result of the movement of electrons in the electric field of this single nucleus. In stationary states of molecules, compared with atoms, due to the presence of several massive nuclei, the dynamic roles of electrons and nuclei change places. Here, not the only heavy nucleus, being "turned off" from the full-fledged dynamics of the entire atomic system, is the "main" source of the electric field that forms the electronic structure of the microsystem. In molecules, such a "main" source of the electric field is the subsystem of light valence electrons, which creates an adiabatic potential that forms the nuclear structure of the microsystem and in which the nuclei make their oscillatory movements. In other words, in the adiabatic approximation, these electrons themselves are "turned off" from the full-fledged dynamics of the entire molecular system. It is this fact of "switching off" in the adiabatic approximation of the valence electron subsystem from fullfledged dynamics that is the condition for the applicability of quantum mechanics to molecular systems. Going beyond the adiabatic approximation leads to the need to consider the full-fledged dynamics of not only nuclei, but also valence electrons. However, due to the incomparability of their masses, in the framework of quantum mechanics this joint dynamics, as shown by the author on the example of elementary electron transfers in condensed matter, becomes singular. In other words, in molecular physics, quantum mechanics works only within the framework of the adiabatic approximation and ceases to work when it goes beyond its framework. The same statement applies to chemical physics, which studies the dynamics of elementary chemical reactions, or in the simplest case, the dynamics of elementary electron transfers in condensed matter, which we consider in this article.

The aforementioned singular dynamics arises, for example, in the process of transitions between the adiabatic ground and first excited states of molecules, as a result of "an attempt by light electrons to reorganize the structure of a very heavy nuclear subsystem only within the

Consider the issue in more detail. So, the singularity in the rate



Fig. 5. Distribution function of black light $\phi(\lambda, T)$ (Eq. (165)).

framework of quantum mechanics", so that the structure of the nuclear subsystem would adapt to a new electron charge distribution in an excited adiabatic (or nearly adiabatic) state. The presence of a singularity in the rates of quantum transitions in molecular quantum mechanics is also indicated by the internal inconsistency of the Franck-Condon principle, which was discussed above in Section 2.5 and which is associated with the inability to give a mathematical description of the qualitative. and essentially quantum-classical, Franck-Condon picture of the dynamics of a molecular transient state. Therefore, in theoretical physics, the discussed singularity must be damped, which was done by the author on the simplest example of elementary electron transfers in condensed matter [10, 11, 12] by introducing chaos (dozy chaos) into the quantum dynamics of the transient state. However, as a result of such damping by introducing chaos, the energy spectrum of moving electrons and nuclei in the transient state becomes continuous [16, 18]. This means that the proposed new dynamic theory is no longer quantum mechanics, but quantum-classical mechanics. Quantum states are the initial and final states in the adiabatic approximation, which are quite sharply different in the electron-nuclear structure, and the transient state is similar to the classical state. The discovery of a new dynamic theory quantum-classical mechanics, means the discovery in physics of a new fundamental property of an electron. This property appears in an electron as a result of its binding of atomic nuclei into molecular systems and consists in its ability to provoke the necessary chaos in the transient dynamic state of these systems. This new property of an electron makes it possible to eliminate the discussed singularity in the quantum mechanics of molecules and, as a result, to ensure the very possibility of "quantum" transitions in molecular systems. Thus, an electron, being exclusively a quantum microparticle in atoms, where it makes quantum jumps between discrete energy levels, in molecules and condensed matter it becomes a quantum-classical microparticle with a continuous spectrum of energy in the transient state [15, 16, 18, 35, 36, 48, 49].¹⁹

2.7. Dozy chaos and quanta: analogy being in their discovery in physics and perhaps in life

It follows from the previous sections that dozy chaos is introduced into theoretical physics as a new physical substance for eliminating the singularity in the rate constants of quantum transitions which occurs in molecular quantum mechanics when going beyond the adiabatic approximation. In theoretical physics, a similar procedure for introducing a new physical substance, namely, quanta, was carried out at the very beginning of the last century to eliminate the singularity in the distribution function of blackbody radiation at high frequencies. As is known from the history of physics, this singularity is often referred to as an ultraviolet catastrophe. Let me remind you the reason for ultraviolet catastrophe. Classical physics assumes that each mode of blackbody radiation, regardless of its frequency, has on average the same amount of energy, namely energy (by $k_{\rm B}T/2$ for the electric and magnetic wave energy). In fact, the average energy per mode \bar{e} depends on its frequency Ω and decreases exponentially with increasing Ω :

$$\overline{\varepsilon} = \frac{\hbar\Omega}{\exp(\hbar\Omega/k_{\rm B}T) - 1}, \quad (\hbar \to 0, \quad \overline{\varepsilon} = k_{\rm B}T).$$
(164)

This fact leads to the elimination of the singularity in the distribution function of black light

$$f(\Omega, T) = \frac{\hbar\Omega^3}{4\pi^2 c^2} \frac{1}{\exp(\hbar\Omega/k_{\rm B}T) - 1}, \quad \phi(\lambda, T) = \frac{2\pi c}{\lambda^2} f\left(\frac{2\pi c}{\lambda}, T\right)$$
(165)

(see Fig. 5) and, as first shown by Planck [95], is a consequence of the fact that the absorption and emission of energy of electromagnetic waves is not continuous, but occurs in the form of quanta $\varepsilon = \hbar \Omega$.

The hypothesis of quanta was conflicting to classical mechanics, and in due time, it was entirely beyond the conventional physical concept, but implications of the discovery of quanta proved to be truly breathtaking. As a matter of fact we can say that quantum mechanics, originating in the hypothesis of quanta, underlies the modern science and technological progress. The quantum-classical mechanics under discussion, based on the hypothesis of dozy chaos, provides an understanding of a series of experimental results in chemistry (see Section 1), which for a long time could not be explained within the framework of the standard quantum mechanics of electron-nuclear motion. From general physical considerations it follows that dozy chaos is the cause not only of molecular quantum transitions, but also of the whole variety of chemical reactions, and as a consequence, it is the physical origin of the evolution of molecular matter, up to the emergence of living matter and the person himself [32b, 33a, 33b]. Therefore, there is every reason to believe that the implications of the discovery of dozy chaos and quantum-classical mechanics in our life will be no less impressive in comparison with the implications of the discovery of quanta and quantum mechanics. From

¹⁹ We are only talking about the outer electrons that are involved in the creation of chemical bonds between atoms. Internal electrons, like in atoms, are tightly bound by their single nucleus.

general considerations it follows, for example, that dozy chaos constitutes the physical basis for the effective functioning of the brain, which is the main "receptacle" of dozy chaos in the living organism [32b, 33a, 33b]. Since the onset of cancer is usually associated with damage to the genes in cells, for example, by radiation, then the radiation of dozy chaos of the living brain in its pathological functioning, for example under stress, is one of the main physical causes of the natural onset and development of cancer [32b, 33a, 33b].

2.8. Outlook for the account of local vibrations in quantum-classical mechanics

The physical picture of molecular quantum transitions (MQT), which was previously discussed, does not at all exclude the presence of quantum jumps in general in the entire dynamics of MQT, in spite of the fact that these quantum jumps are absent in our theory of elementary electroncharge transfers [36, 96]. In our theory, this absence is due to taking into account only the simplest dynamics of MQT, namely, the dynamics that is associated with the interaction of the electronic transition with non-local phonons of the environment; a more complex dynamics of the interaction of the electronic transition with local vibrations is neglected there.²⁰ Taking the latter into account will lead to the account in theory, along with the chaotic dynamics of the transient state of electron-vibrational transitions, as well as vibrational quantum jumps. When, together with the reorganization of the nuclear subsystem, the change in the electronic state during the electron transition to the final state is already completed, but in the molecular system there is still some small energy defect, determined by the energy conservation law, then there will be a rapid quantum jump in the vibrational energy of the nuclei, that being already in the final adiabatic electronic state (more precisely, in a state close to adiabatic, if we are talking about the optical excitation of the molecule), in order to eliminate this energy defect. In other words, the quantum jump or transition in the local vibrational mode in the entire process of a molecular "quantum" transition occurs at its final stage, namely, at the moment of transition of a molecule or molecular system from chaotic (dozy-chaos) dynamics to adiabatic (or nearly adiabatic) dynamics. In the experiment, this effect is manifested in the presence of striped spectra and discrete narrow lines in the electron-vibrational spectra that are well-known from the literature (see Refs. [62, 63, 64]).

Within the framework of quantum-classical mechanics, the wellknown small line widths corresponding to local oscillations can easily be justified by an estimate that uses our result for the shape of the optical absorption band corresponding to the case of non-local oscillations (see Eqs. (141), (142), (143), (144), (145), (146), (147), (148), (149), (150), (151), (152), (153), (154), (155), (156), (157), and (158) in Section 2.4.8). We assume that the shape of the optical line for local oscillations is close to the Gaussian form known from the standard theory [72], and which in our theory corresponds to the case when the dozy-chaos energy γ is equal to or of the order of the reorganization energy *E*: $\gamma \cong E$. In the standard theory [72], the half-width of the optical line (see Eq. (163) in Section 2.4.9) is determined, in particular, by the reorganization energy $\lambda_r \equiv 2E$ and is calculated from the following equation

$$w_{1/2} = 2\sqrt{2\ln 2}\sqrt{2\lambda_{\rm r}k_{\rm B}T}.$$
 (166)

It follows from Eq. (37), which determines the energy of reorganization $E = \frac{1}{2} \sum_i \hbar \omega_i \tilde{p}_i^2$, that, other things being equal, its value for one nonlocal or local mode of oscillations of nuclei is much less than its value for a large number of the modes of oscillations of nuclei interacting effectively with the electronic transition. From this and from Eq. (166), small line widths follow, which must be obtained within the framework of quantum-classical mechanics in the future in the case when only local oscillations of the nuclei are taken into account.

Thus, in the general case (that is, taking local vibrations into account) in our picture, just as in the Franck-Condon picture, MQT occur in two stages. However, while in the Franck-Condon picture the rapid quantum jump (in the electron energy) precedes the slow process of nuclear reorganization, in our picture, on the contrary, the quantum jump (in the vibrational energy of the nuclei) occurs when the reorganization of the nuclei, as a result of the electronic transition, has already been completed. The construction of an adequate mathematical technique for the accurate description of such a physical picture is a separate and rather complex problem, the solution of which will require the collective efforts of the scientific community in the future. Until this solution is obtained, one can continue to use successfully the standard and essentially erroneous Franck-Condon explanation given in the existing literature (see Refs. [62, 63, 64]).²¹

The elementary transfer of electron charge in condensed media, when in theory only non-local phonons are taken into account, traditionally refers to chemical physics. In other words, in this article we have demonstrated, strictly speaking, only the fact that quantum mechanics stops to operate in chemical physics and the need to introduce chaos (dozy chaos) in the transient state of elementary electron transfers. In molecular physics, where local phonons must be taken into consideration in theory, this demonstration, as previously mentioned, has yet to be realized. However, more recently, the emergence of chaos in the problem of excitation of a hydrogen molecule under the influence of a periodic external force was theoretically proved [97].

2.9. Quantum-classical mechanics as non-mesoscopic physics

Mesoscopic systems are those that are in between the quantum and classical mechanical states. Therefore, the question arises: "How do quantum-classical mechanics reconcile with mesoscopic physics?"

Mesoscopic physics is a phenomenological approach, that is, such an approach when we build *ad hoc* models to account for quantum effects in macroscopic and classical systems. For example, when we consider dimensional quantization (quantum confinement) in nanoparticles, or when we introduce quantization of classical chaos, we obtain quantum chaos. Various examples of mesoscopic physics problems can be found, for example, in Ref. [98], and, in particular, on quantum chaos and its connection with mesoscopic physics, for example, in Ref. [99].

The author's approach is a movement in the opposite direction: this is when we are trying to solve honestly *a priori* quantum problem for a fairly complex physical system.²² In contrast to the phenomenological approach in mesoscopic physics, our approach can be called a microscopic approach. An example of such a microscopic approach is the

²⁰ For non-local vibrations of nuclei (crystal lattice phonons), the manyphonon transitions with phonon occupation numbers changing by over 1 do not contribute to the result for the rate constant of quantum transitions and can be neglected. In the case of localized phonons (local vibrations), the manyphonon transitions with all phonon occupation numbers varying from zero "ad infinitum" make the contribution to the result and hence many of them must be taken into account. About the difference in the interaction of the electronic transition with non-local and local phonons, you can see Section 2.4.7 and a review by Perlin [72].

²¹ Nevertheless, the erroneous Franck-Condon picture for simulation of the transient state of MQT can continue to be used even after the correct solution of the MQT problem will also be obtained in the general case taking local vibrations into account. For example, in our daily life we successfully use the erroneous idea that every day the Sun rises in the East and sets in the West, although in reality it is not about the motion of the Sun in the sky, but the rotation of the Earth around its axis. However, when we go beyond the surface of the Earth into Space, the use of the correct physical picture becomes absolutely necessary. In the simplest case of MQT, this "entry into Space" corresponds to the construction of the theory of the shape of the optical bands in polymethine dyes and their aggregates (see Section 1).

²² An a posteriori quantum-classical problem.

standard theory of many-phonon processes [72], which describes quantum transitions and spectra of luminescence and optical absorption in F-centers [73, 86, 93]. In this theory, the initial and final states of the system are considered in the adiabatic approximation. Because of using the adiabatic approximation there, it is possible to obtain a result for the rate constants of quantum transitions in the framework of quantum mechanics, which contains a dependence on temperature as a consequence of averaging over the equilibrium (Planck) distribution of the initial states in the "molecular" ensemble, in particular, in the ensemble of F-centers. The reason for the success of this approach in explaining optical bands in F-centers is determined by the fact that the standard theory of many-phonon transitions [72] is essentially a special case of quantum-classical mechanics, the case for strong dozy chaos (see Sections 2.4.9 and 2.5). An attempt to go beyond the adiabatic approximation in quantum mechanics, in the theory of many-phonon transitions, leads to a singularity in the rates of quantum transitions. Expressed in a different way, going beyond the adiabatic approximation in quantum mechanics automatically leads to the theory of many-phonon transitions beyond quantum mechanics. Solving this situation is allowed by the natural damping of the singularity, which is described in detail in Section 2.4. Although this natural and forced damping of the singularity does not return the theory of many-phonon transitions to the realm of quantum mechanics, it does provide a physical (non-singular) result for the optical spectra (Section 2.4.8), and a much more general one than given by the standard theory of many-phonon transitions (Figs. 1, 2, and 4b). The reason for going beyond quantum mechanics is, ultimately, to take into account the dynamics of the transient state, which, as it turns out, is chaotic. This chaos in the motion of optical electrons and associated atomic nuclei in the transient state leads to a continuous energy spectrum of the entire "electron + nuclear environment" system for a short time (order \hbar/E) of the transition between its initial and final states. That is to say, for a short transition time, the "electron + nuclear environment" system becomes similar to the classical system. In this case, the initial and final stationary states continue to be quantum states, that is, such states that can be described with sufficient success in the adiabatic approximation. Hence the terminology "quantum-classical mechanics."

The question can be raised whether the ensemble and with it the temperature, which, as a rule, determine the experimental reality, brings an element of classicality into the quantum system, regardless of whether the chaotic dynamics of the transient state in individual molecules is taken into account. The answer is yes, because in the standard theory of many-phonon transitions, we already obtain the result for the shape of the optical band as a continuous function (as a result of the dispersion of phonon frequencies [72]), rather than discrete lines, as in atomic physics. In this experiment, we have the same shape of the optical band as a continuous function. However, by applying the adiabatic approximation, this result for the shape of the optical band as a continuous function in the standard theory of many-phonon transitions is obtained completely within the framework of quantum mechanics. This fact allows us to consider many-phonon transitions in the standard theory as "purely" quantum transitions, and not as quantum-classical transitions, as we do in the new theory, where classicality already arises in the transient state of each single molecule (each donor-acceptor couple in the case of electron transfers) in the ensemble, and to consider the most standard theory of many-phonon transitions not as a theory belonging to mesoscopic physics, but as a quantum-mechanical theory. Concomitantly, it is essential that taking into account only the ensemble of molecules and the associated temperature to simulate an experimental situation, without introducing chaos into the transient state of each single molecule, does not solve the problem associated with the appearance of a singularity in the transition rates in the case of going beyond the adiabatic approximation. In short, classicality, which is introduced into quantum mechanics by an ensemble of molecules and temperature, is not capable of damping the essential

singularity in the transition rates.

The physical nature of the singularity is discussed in this article in several aspects: 1) the incomparability of the masses of electrons and nuclei in the presence of full-fledged dynamics of both nuclei and electrons (Sections 2.1 and 2.4.3); 2) quantum transitions in a potential box with a moveable wall (Section 2.6); and 3) internal inconsistency of the Franck-Condon principle (Section 2.5). For example, consider the aspect due to the internal inconsistency of the Franck-Condon principle. This internal inconsistency is associated with the impossibility in the framework of quantum mechanics to formalize the qualitative picture of the transition proposed by the Franck-Condon principle, which assumes that in the adiabatic approximation, the same oscillator behaves like a classical and quantum system in the same transition act. Namely, according to the Franck-Condon principle, in the oscillator there is a classical movement of the nuclei to the turning point at which the quantum transition to a new electronic state occurs. Therefore, the Franck-Condon principle with its quantum-classical paradigm is a precursor to the creation of quantum-classical mechanics with its dozy chaos in the transient state of an individual molecule. In the new theory, which gives new physical results in the cases of weak and medium dozy chaos, the Franck-Condon principle acts as an effective simulator of strong dozy chaos. Thus, the new theory of quantum-classical mechanics is not another section of mesoscopic physics with its phenomenological approach, but a natural generalization of the quantum-mechanical theory of manyphonon transitions [72], where the dozy-chaos energy $\gamma \equiv 0$,²³ in the case of $\gamma \neq 0$, that is, taking into account the chaotic dynamics of the transient state in elementary electron transfers in condensed matter, and also in molecules [65] (see Section 2.8) and supramolecular systems in the future. This is a generalization similar to the broad meaning in which Schrödinger's equation as a postulate in quantum mechanics is a generalization of the Hamilton-Jacobi equation for action in classical mechanics, and in general, quantum mechanics is a generalization of classical mechanics, where Planck's constant $\hbar \equiv 0$, in the case of $\hbar \neq 0$ (see Section 2.7). Comparisons between quantum-classical mechanics and other theoretical approaches, where mixing quantum and classical mechanics is undertaken (see e.g. [100, 101, 102]), will be given elsewhere.24

2.10. Dozy chaos as compared to quantum chaos

In modern mesoscopic physics (see e.g. Refs. [98, 99]), there is an idea of the so-called quantum chaos, which has been actively studied, mainly theoretically, for several decades now (see Refs. [103, 104, 105, 106, 107, 108, 109, 110, 111, 112, 113, 114, 115, 116, 117, 118, 119]). Dozy chaos differs fundamentally in physical nature from quantum chaos. The term "quantum chaos" is generally understood to comprise all problems concerning the quantum mechanical behavior of classically chaotic systems [103, 104, 105, 106, 107, 108, 109, 110, 111, 112, 113, 114, 115, 116, 117, 118, 119]. In other words, systems, whose underlying classical dynamics is chaotic due to nonlinear interactions [120, 121, 122, 123], exhibit signatures of the chaos in their quantum mechanics. The quantization of a classical dynamical system with chaos is a restriction on chaos, and hence this quantization leads only to a less pronounced (inferior) chaos. Therefore, Berry [104, 105] urged at one time to speak not of quantum chaos, but of quantum chaology, and Chirikov [107] used the term "quantum pseudochaos". Nevertheless, the term "quantum chaos" has remained generally accepted (see e.g. Refs. [103, 108]). One of remarkable results of the theory of quantum chaos are the Heller's scars [106].

As for dozy chaos, in the case of classical mechanics it simply ceases to exist, since in this case molecular quantum transitions, the dynamics of

 $^{^{23}}$ More precisely, in quantum mechanics energy γ is an infinitely small value.

²⁴ As a rule, these theoretical approaches do not demonstrate clear applications to the experiment.

which dozy chaos determines only, cease to exist. Hence, dozy chaos, which provides a classical character of any transient state in molecular quantum transitions, is a purely quantum phenomenon in physical nature, in the sense that it does not exist by definition, in contrast to quantum chaos originating from classical chaos, in the field of classical phenomena. Such a dualistic nature of dozy chaos — its existence only in the field of quantum phenomena and the functional role being in providing, due to chaos, a classical character of any transient state in molecular quantum transitions — is a generalization (in the case of energy $\gamma \neq 0$) of the wave-particle dualism in quantum mechanics ($\gamma = 0$), which is associated with the statistical nature of physical meaning of the wave function.

Dozy chaos, as compared to quantum chaos, is not associated with nonlinear interactions in a molecular system: dozy chaos appears in a linear problem (see the linear term $\sum_i U_i(\mathbf{q})p_i$ responsible for the electron-phonon interaction in the Hamiltonian in Eq. (19)). In addition, in contrast to the "inferiority" of quantum chaos, because of the classical nature of the transient state of molecular "quantum" transitions, dozy chaos, like classical chaos, is a full-fledged chaos.

Thus, while the analogue of quantum chaos is classical chaos, there is no analogue for dozy chaos in the field of classical phenomena. As previously mentioned, dozy chaos is the original and universal physical substance, the universal to the same extent to which the electrons and nuclei themselves, the dynamic interactions of which it provides in any "quantum" transition from one of their bound state to another, are universal.

As is clear from the foregoing, the concept of quantum chaos is constructed, by definition, so as to preserve the standard paradigm of quantum mechanics and not to seek a review of its basis in connection with the inclusion of chaos in quantum dynamics. Vice versa, the concept of dozy chaos is constructed, by definition, in such a way that requires a revision of the basis of quantum mechanics, to incorporate the chaotic and classical in nature dynamics into the transient state of molecular quantum transitions.

Summarizing, we can say that the emergence of quantum chaos, for example, in elementary chemical processes, is theoretically considered within the framework of quantum mechanics [112, 113, 114, 115, 116, 117, 118, 119], and quantum chaos becomes a classical chaos when the transition from quantum mechanics to classical mechanics occurs. (Note that in this case of elementary chemical processes with quantum chaos, no clear explanations for or comparisons with the experiment were discussed.) In other words, quantum chaos in quantum mechanics is some analog of classical chaos in classical mechanics. In contrast, there is no analogue of dozy chaos, which is associated with the specific property of an electron to provoke chaos in the vibrations of nuclei, in classical mechanics. In other words, dozy chaos is a "purely quantum" phenomenon related to the specific property of a dynamically active electron in a molecule. We also note that the emergence of quantum and/or classical chaos is associated, as is well-known, with some nonlinear interactions in physical systems, whereas the emergence of dozy chaos occurs in a system with a linear interaction. More detailed comparisons between dozy chaos and quantum chaos will be given elsewhere.

2.11. Mathematical techniques and perspectives in quantum-classical mechanics

The formal mathematical techniques are discussed in this section. It is well-known that the standard theory of many-phonon transitions [72] extensively uses the following techniques [11, 12, 14, 28, 79]: the generating polynomial (generating function) technique of Krivoglaz and Pekar [76, 77], which is similar to the Darwin-Fowler method [72, 78] from statistical physics (see Section 2.4.3, footnote 9), the operation calculus of Feynman and Lax [94, 124], the density matrix technique of Kubo and Toyozawa [125, 126], and the quantum field theory technique (see Ref. [72]). Among all methods, the simplest and physically transparent method is the technique of generating functions of Krivoglaz and

Pekar. The simplicity and transparency of this technique is due to the fact that it is formulated in terms of the wave function. However, this technique takes into account only the main effect of the electron-phonon interaction - shifts of normal phonon coordinates. The change in the phonon frequencies during electronic transitions and other less significant effects can be taken into account on the basis of the remaining aforementioned techniques. Since the processes of elementary electron-charge transfers in condensed matter and other electron-phonon transitions, in which the dynamics of the transient state must be taken into account, are much more complicated than the standard case of electron-phonon transitions, it is obvious that a generalization of the standard theory of many-phonon transitions should be started, using the simplest technique of Krivoglaz and Pekar. In my technique [11, 12, 52, 75, 79] that generalizing the technique of Krivoglaz and Pekar, as previously shown (Section 2.4.6), all of the intermediate electron-phonon states arising during elementary electron-charge transfers are precisely taken into account. The use of a more sophisticated formal technique, for example, the technique of Feynman path integrals [127], would greatly complicate the whole problem rather than simplify it. I believe that the next main stage in the development of the quantum-classical mechanics of elementary electron transfers [11, 12, 15, 79] will be based on a generalization of the Kubo-Toyozawa density-matrix technique or the quantum field theory technique.

2.12. Quantum-classical mechanics and quantum information

This section discusses once again the role of dozy chaos in the elimination of the aforementioned singularity in molecular quantum transitions [15, 34]. In this section, we want to emphasize the fact, which is associated with the elimination of this singularity, that the chaotic nature of the interaction of very light electrons and heavy nuclei in the transient state of molecular "quantum" transitions is the reason for their self-organization. As a result of electron provocation of chaos in the vibrational motion of nuclei, part of the vibrational motion is transformed into the translational movement of their equilibrium positions to new positions that correspond to a new distribution of electron charge. In the process of completing the formation of this new distribution of the electron charge in the final state, the resulting translational motion of the nuclei is transformed back into their vibrational motion, but near the new equilibrium positions.

Thus, as the driver of the reorganization process of the nuclear subsystem in a molecule in addition to the electronic motion, which by creating chaos starts the process, the very vibrational motion of the nuclei acts, which is transformed into a chaotic motion, performing in effect the very process of reorganization of the nuclei. (Vibrations of the nuclei in a molecule are always present, even at absolute zero and zeropoint oscillations of the nuclei.) Moreover, the motion of the electrons and nuclei in a molecule in the process of a quantum transition (for example, excitation of an electron-vibrational transition) is undergoing a dramatic change in its nature. Namely, the quantum motion in the adiabatic ground state dies as a result of the launch of the chaotic process of a "quantum" transition and again revives as a result of the completion of this process in the excited adiabatic (or nearly adiabatic) state.²⁵ The motion of the electrons and nuclei in the transient state of a molecule has a paradoxical classical nature at first glance.

Above, we described a new mechanism of electron-vibrational transitions as if the electron has free will. In fact, we are discussing the dynamic self-organization of complex molecular systems, which is presented here on the simplest example of the dynamic self-organization of electron-vibrational transitions in molecules. Similar issues are widely

²⁵ The remark in parentheses does not apply, for example, to elementary electron-charge transfers or elementary chemical reactions, in which "quantum" transitions from one adiabatic ground state to another adiabatic ground state take place.

discussed at present in such fields of physics as quantum information and cybernetic physics (see Refs. [128, 129, 130, 131, 132, 133, 134, 135, 136, 137, 138, 139, 140] and references therein). In a chaotic molecular transient state, there is essentially an intense exchange of information between a very light electron and heavy nuclei about their current energies and states of motion. The consequence of this exchange of information is the regular, but not singular, dynamics of the transient state [14, 28].

3. Conclusions

This article focuses on the regular exposition of dozy-chaos mechanics or quantum-classical mechanics of elementary electron transfers in condensed matter, which are the simplest cases of molecular quantum transitions. For the theoretical description of molecular quantum transitions, quantum-classical mechanics arises in molecular and chemical physics instead of quantum mechanics as a consequence of a critical examination of the Franck-Condon principle, widely known and popular in optical molecular spectroscopy, which was introduced in optical spectroscopy as an external addition to quantum mechanics almost 100 years ago. Quantum-classical mechanics allows, for example, carrying out a theoretical examination of the shapes of optical bands in polymethine dyes and their aggregates, which can only be explained if the Franck-Condon principle is abandoned in principle. According to the arguments given in this article, the Franck-Condon principle remains to date an unsubstantiated hypothesis, despite its successful application to explain a substantial number of experimental facts. The dynamics of the molecular quantum transition, which is given by the Franck-Condon principle, does not stand up to criticism on the basis of general physical considerations (see Sections 2.1 and 2.5). The main critical circumstance is the fact that up to the present, in the framework of quantum mechanics, there is no distinct quantitative solution to the problem of the dynamics of the molecular transient state, at least in some simplest cases, which would confirm the Franck-Condon principle and/or the Franck-Condon dynamics of the transient state. Moreover, the study of the dynamics of the molecular transient state has a physical meaning only when going beyond the Born-Oppenheimer adiabatic approximation, since in this approximation, the full-fledged dynamic role of the electronic subsystem is absent, and the role of the electronic subsystem is reduced solely to the formation of the adiabatic potential in which the nuclei move. However, as shown by the author in the example of a new theory of elementary electron transfers in condensed matter, going beyond the adiabatic approximation leads to a singularity in the rate constants of "quantum" transitions (see Section 2.4.3), which agrees with the general qualitative physical considerations in Section 2.1. At a qualitative level, a vivid physical demonstration of this singularity is given in Section 2.6 in the example of a potential box with a moveable wall. Therefore, to correctly consider the dynamics of the molecular transient state, this singularity must be damped. In other words, we are forced to introduce into molecular quantum mechanics some additional assumption or postulate that this damping would provide, similar to how quantum mechanics was in due time forced to be supplemented from the outside by the Franck-Condon principle.

In the example of the problem of elementary electron transfers in condensed matter, the specified singularity is easiest to dampen if the infinitesimal imaginary additive in the energy denominator of the spectral representation of the total Green's function of the system is replaced by its finite value. In other words, for the first time the problem of the dynamics of a molecular transient state was solved quantitatively by the example of the problem of elementary electron transfers in condensed matter, that is, the solution to the problem was found as discussed in Section 2.1. A successful quantitative explanation, based on this new theory, of the basic set of experimental data on the shape of the optical bands in polymethine dyes and their aggregates showed that the absolute value of the new imaginary additive in the energy denominator of the total Green's function of the system significantly exceeds the magnitude

of the quantum of nuclear vibrations. This additive could be interpreted as the energy width of the electron-vibrational virtual levels of the transient state, which provides a multiple exchange of motion and energy between different vibrational modes and the electron in the transient state. However, because of the aforementioned excessively large value of this additive, the exchange of motion and energy between different vibrational modes and the electron in the transient state proves to be so intense that it leads to chaos both in the motion of the medium nuclei and in the motion of the electron that undergoes an elementary transfer. Moreover, from the physical perspective, this can be interpreted in such a way that it is the electron, as very light and mobile, rather than the surrounding nuclei, that provokes chaos in a transient state. The electron provokes chaos to be able to easily control the movement of very heavy nuclei in a transient state. This involves the discovery of a new and universal property of an electron, namely, the property of creating chaos in the motion of nuclei in a transient state during a molecular quantum transition. Electrons acquire this universal property when they bind atomic nuclei to molecules. This is called dozy chaos, since it is not present either in the initial or final states and arises only during molecular quantum transitions. The presence of strong chaos (dozy chaos) in a transient state easily explains why the Franck-Condon principle is often so effective in applications. In strong dozy chaos, as previously shown using the example of elementary electron transfers, the rate constants of molecular quantum transitions cease to depend on the dynamics of the transient state (see Section 2.6, Fig. 4b, where the case of strong dozy chaos corresponds to the blue low-intensity Gaussian-like curve). They depend only on the initial and final adiabatic states of the molecular system, with which the Franck-Condon principle essentially only deals. Apparently, this is why dozy chaos has eluded the attention of researchers for so long.

Since according to the Franck-Condon principle, the same oscillator behaves in the same elementary molecular act both as a quantum system (near the turning points of the nuclei) and as a classical system (near the bottom of the potential well for the nuclei), then the Franck-Condon principle is essentially one of the most important prerequisites for the creation of quantum-classical mechanics (see Sections 2.1 and 2.5), the simplest implementation of which is demonstrated in this article.

When dozy chaos is not strong enough to be simulated by applying the Franck-Condon principle, elements of dynamic self-organization manifest themselves in the chaotic dynamics of the molecular transient state. It is to this case that electronic transitions occur in polymethine dyes and their aggregates in solvents. Taking into account these elements of dynamic self-organization in theory allowed the author to quantitatively explain the optical band shapes in polymethine dyes and their aggregates (Section 1). The dynamic self-organization of the transient state is most clearly manifested in the case of J aggregates, where the elementary transfer of the alternating electron charge along the quasilinear optical chromophore is facilitated by a quasi-synchronous motion of the environmental nuclei (Section 1). This fact manifests itself in a very narrow and intense optical J band, and in this connection J aggregates are widely used in a variety of applications (see Refs. [14, 28] and references therein). An overview of other cases of successful applications of dozy-chaos mechanics is briefly provided in the introduction to this article (Section 1).

Quantum-classical (dozy-chaos) mechanics of the elementary electron transfers in condensed matter, which was earlier used to quantitatively explain the optical band shapes in polymethine dyes and their aggregates (Section 1), allows for only non-local phonons and neglects local phonons. As a consequence, this leads to a relatively simple transient state with a continuous spectrum of electron-vibrational energies, that is, during molecular "quantum" transition — elementary electroncharge transfer, the quantum system — "electron + environment" becomes a classical system. This provides (together with the dispersion of phonon frequencies [72]) the continuous optical spectra of polymethine dyes and their aggregates, which were discussed in Section 2.1. Taking into account local phonons is a much more complicated problem than considering non-local phonons, and the solution will require considerable efforts from researchers in the future. This account will also lead to a complication of the transient state, in which, along with the chaotic classical dynamics of the electron-nuclear motion, quantum jumps in the vibrational energy of the nuclei at the final stage of the molecular "quantum" transition appear, that is, after the reorganization of the entire nuclear subsystem has already been completed (see Section 2.8). As a result of this account, in the theoretical optical spectra, along with continuous bands corresponding to non-local phonons, striped spectra appear corresponding to both non-local and local phonons. The dominance of any particular local modes in the molecular system should lead to well-defined discrete and narrow vibrational lines in the theoretical and experimental optical spectra (see Section 2.8).

The methods for calculating the shape of the optical bands in polymethine dyes and their aggregates, based on the quantum classical mechanics of electron transfer, can easily be extended to a wide range of other similar objects studied in organic chemistry, for example, styryl dyes and charge-transfer complexes. Generalization to more complicated cases of quantum-classical transitions than electron transfer, due to the large cumbersomeness of the mathematical apparatus arising here, will require the collective efforts of the scientific community.

To confirm the new physical picture of molecular quantum transitions, based on the concept of dozy chaos, it is of great interest to carry out experiments on its direct detection [15, 18, 33, 34, 35, 36, 96]. In this regard, it would be interesting, for example, to register the loss of regularity of the structure of a molecule in its chaotic transient state [96], using modern X-ray free electron lasers [141].

In conclusion, future research should focus on quantum-classical mechanics and its application to luminescence spectra of J aggregates and their comparison with absorption spectra [142].

Declarations

Author contribution statement

All authors listed have significantly contributed to the development and the writing of this article.

Funding statement

This work was supported by the Ministry of Science and Higher Education within the State assignment Federal Scientific Research Center "Crystallography and Photonics" Russian Academy of Sciences.

Competing interest statement

The authors declare no conflict of interest.

Additional information

Data associated with this study has been deposited at Dryad Digital Repository under the accession number https://doi.org/10.5061/dr yad.t0r3p.

Acknowledgements

This article is dedicated to my daughter Olga and granddaughters Ekaterina and Maria.

I thank all of the participants in the European XFEL Theory Seminar (March 6, 2018, Schenefeld, Hamburg, Germany), especially Serguei L. Molodtsov, Ruslan P. Kurta, and Alexander I. Lichtenstein, for their useful discussion of the main results of this article.

References

- R. Brout, F. Englert, E. Gunzig, The origin of the Universe as a quantum phenomenon, Ann. Phys. (N. Y.) 115 (1978) 78–106.
- [2] D. Atkatz, H. Pagels, Origin of the Universe as a quantum tunneling effect, Phys. Rev. D 25 (1982) 2065–2073.
- [3] J.B. Hartle, S.W. Hawking, Wave function of the universe, Phys. Rev. D 28 (1983) 2960–2975.
- [4] R. Penrose, Fashion, Faith, and Fantasy in the New Physics of the Universe, Princeton University Press, Princeton, 2016.
- [5] R. Penrose, The Emperor's New Mind: Concerning Computers, Minds, and the Laws of Physics, Oxford University Press, Oxford, 1989.
- [6] R. Penrose, Shadows of the Mind: an Approach to the Missing Science of Consciousness, Oxford University Press, Oxford, 1994.
- [7] M.P.A. Fisher, Quantum cognition: the possibility of processing with nuclear spins in the brain, Ann. Phys. (N. Y.) 362 (2015) 593–602.
- [8] H. Atmanspacher, Quantum Approaches to Consciousness, Stanford Encyclopedia of Philosophy, Stanford University, Stanford, CA, 2015.
- [9] R. Penrose, S. Hameroff, S. Kak (Eds.), Consciousness and the Universe: Quantum Physics, Evolution, Brain & Mind, Cosmology Science Publishers, Cambridge, MA, 2017.
- [10] V.V. Egorov, Electron-transfer approach to the nature of the optical lineshape for molecular J-aggregates, Chem. Phys. Lett. 336 (2001) 284–291.
- [11] V.V. Egorov, On electrodynamics of extended multiphonon transitions and nature of the J-band, Chem. Phys. 269 (2001) 251–283.
- [12] V.V. Egorov, Nature of the optical transition in polymethine dyes and Jaggregates, J. Chem. Phys. 116 (2002) 3090–3103.
- [13] V.V. Egorov, Dynamic pumping of elementary charge transfer by environmental dissipative reorganization, Russ. J. Electrochem. 39 (2003) 86–96.
- [14] V.V. Egorov, M.V. Alfimov, Theory of the J-band: from the Frenkel exciton to charge transfer, Phys. Uspekhi 50 (2007) 985–1029.
- [15] V.V. Egorov, Optical lineshapes for dimers of polymethine dyes: dozy-chaos theory of quantum transitions and Frenkel exciton effect, RSC Adv. 3 (2013) 4598–4609.
- [16] V.V. Egorov, Nature of the narrow optical band in H*-aggregates: dozychaos-exciton coupling, AIP Adv. 4 (2014) 077111-1-077111-9.
- [17] A. Petrenko, M. Stein, Molecular reorganization energy as a key determinant of Jband formation in J-aggregates of polymethine dyes, J. Phys. Chem. A 119 (2015) 6773–6780.
- [18] V.V. Egorov, Nature of the optical band shapes in polymethine dyes and H-aggregates: dozy chaos and excitons. Comparison with dimers, H*- and J-aggregates, R. Soc. Open Sci. 4 (2017) 160550-1–160550-20.
- [19] V.V. Egorov, Data from: Dryad digital repository, R. Soc. Open Sci. 4 (2017) 160550-1–160550-20.
- [20] V.V. Egorov, On electron transfer in Langmuir-Blodgett films, Thin Solid Films 284–285 (1996) 932–935. Erratum, *Thin Solid Films* 299, 1997, 190 [Proc. 7th Int. Conf. on Organized Molecular Films, Ancona, Italy, 10–15 September 1995 (Elsevier, Amsterdam, 1996)].
- [21] V.V. Egorov, Electron transfer in condensed media: failure of the Born-Oppenheimer and Franck-Condon approximations, collective phenomena and detailed balance relationship, J. Mol. Struct.: THEOCHEM 398–399 (1997) 121–127 [Proc. 4th World Congr. Theor. Oriented Chemists – WATOC '96, Jerusalem, Israel, 7–12 July 1996 (Elsevier, Amsterdam, 1996)].
- [22] V.V. Egorov, in: 4th World Congr. Theor. Oriented Chemists WATOC '96, Jerusalem, Israel, 7–12 July, 1996. Book of Abstracts.a)Electron Transfer in Condensed media: Failure of the Born-Oppenheimer and Franck-Condon Approximations, and Collective Phenomena, p. 107; b)Theory of Elementary Charge-Transfer Processes in Acid-Base Catalysis, p. 167; c)Dispersion of the Rate of Electron Transfer from Excited Electronic States of Aggregated Molecules, p. 168; d)Axiomatic Noncompleteness of Quantum Mechanics for Charge Transfer in

d)Axiomatic Noncompleteness of Quantum Mechanics for Charge Transfer in Condensed Matter, p. 172.

- [23] V.V. Egorov, Tunnel luminescence: failure of the Born-Oppenheimer and Franck-Condon approximations and collective phenomena, J. Lumin. 72–74 (1997) 871–873 [Proc. Int. Conf. Lumin. Opt. Spectr. Cond. Matter – ICL '96, Prague, Czech Republic, 18–23 August 1996 (Elsevier, Amsterdam, 1996)].
- [24] V.V. Egorov, The superexchange through virtual phonons in the dynamics of elementary electron transfer from excited electronic states of aggregated molecules, J. Lumin. 76–77 (1998) 544–547 [Proc. 11th Int. Conf. on Dynamical Processes in Excited States of Solids – DPC '97, Mittelberg, Kleinwalsertal, Austria/Germany, 20–24 July 1997].
- [25] Chimia 51 V.V. Egorov, in: Nr. 7 [Papers Presented at the Frontiers in Chemistry, New Perspectives for the 2000s: 36th IUPAC Congress, Geneva, Switzerland, 17–22 August 1997], 1997.a)The Dynamics of Elementary Electron Transfer in Condensed Matter: Superexchange through Phonons, p. 554, AC-P2; b)The Dynamics of Elementary Proton-Transfer Processes, p. 555, AC-P3; c)The Dynamics of Elementary Electron Transfer from Excited Electronic States of Aggregated Molecules, p. 555, AC-P4;
 d) The Real and Virtual Interplay between the Electron and Nuclear Motion in the
- Dynamics of an Elementary Electron Transfer in Condensed Matter, p. 555, AC-P5. [26] V.V. Egorov, Self-coherency of Extended Electron Transitions, a) , in: 12th Int.
- Conf. On Dynamical Processes in Excited States of Solids DPC '99, Humacao, Puerto Rico, USA, 23–27 May, 1999, p. 1. Abstract Book;
 b) , in: 9th Int. Conf. On Unconventional Photoactive Systems, Wuerzburg, Germany, 31 August–4 September, 1999, p. 55. Abstract Book.

- [27] V.V. Egorov, Nature of the Optical Transition in Polymethine Dyes and J-Aggregates, in: XX Int. Conf. On Photochemistry, Moscow, Russia, 30 July–4 August, 2001, pp. 232–233. Abstract Book.
- [28] V.V. Egorov, Theory of the J-band: from the Frenkel exciton to charge transfer, Phys. Proc. 2 (2009) 223–326 [Proc. 15th Int. Conf. Lumin. Opt. Spectr. Cond. Matter – ICL '2008, Lyon, France, 7–11 July 2008].
- [29] V.V. Egorov, Optical line shapes for polymethine dyes and their aggregates: novel theory of quantum transitions and its correlation with experiment, J. Lumin. 131 (2011) 543–547 [Proc. 17th Int. Conf. on Dynamical Processes in Excited States of Solids (DPC'10), Argonne Nat. Lab., Argonne, Illinois, USA, 20–25 June 2010].
- [30] V.V. Egorov, Nature of the J- and H*-bands: Dozy Chaos and Excitons, in: 16th Int. Conf. Lumin. Opt. Spectr. Cond. Matter (ICL'11), Ann Arbor, Michigan, USA, 26 June–1 July, 2011. Book of Abstracts, WV5.
- [31] V.V. Egorov, Novel theory of charge transfers in condensed matter and its correlation with experiment: optical line shapes for polymethine dyes and their aggregates, in: ESF Conf. – Charge Transfer in Biosystems, Obergurgl, Austria, 17–22 July, 2011. http://psi-k.net/download/workshops/2011/Obergurgl_ Jul11.pdf.
- [32] V.V. Egorov, in: 43rd IUPAC world chemistry congress, San Juan, Puerto Rico, USA, 31 July–5 August, 2011. https://ru.scribd.com/document/94547645/lupac 2011-Book.a)Optical Line Shapes for Polymethine Dyes and Their Aggregates: Novel Theory of Quantum Transitions and its Correlation with experiment, Book of Abstracts (1149), p. 489; b)Discovery of Dozy Chaos and Discovery of Quanta: Analogy Being in Science

and Perhaps in History of Civilization, Book of Abstracts (1266), p. 523.

- [33] V.V. Egorov, Discovery of Dozy Chaos and Discovery of Quanta: Analogy Being in Science and Perhaps in Human Progress, a), in: 243rd ACS national meeting & exposition, San Diego, California, USA, 25–29 March, 2012. Book of abstracts, 282, http://acselb-529643017.us-west-2.elb.amazonaws.com/chem/243nm/p rogram/view.php?pub_num=282&par=BIOL;
 b) S.G. Stavrinides, S. Banerjee, H. Caglar, M. Ozer (Eds.), Chaos and Complex Systems: Proc. 4th Int. Interdisciplinary Chaos Symp., Springer, Berlin, 2013, pp. 41–46.
- [34] V.V. Egorov, Dozy chaos in chemistry: simplicity in complexity, in: S.G. Stavrinides, S. Banerjee, H. Caglar, M. Ozer (Eds.), Chaos and Complex Systems: Proc. 4th Int. Interdisciplinary Chaos Symp., Springer, Berlin, 2013, pp. 219–224.
- [35] V.V. Egorov, Dozy-chaos nature of molecular quantum transitions and theory of the optical band shapes in polymethine dyes, 2nd Int. Conf. on Atomic and Nuclear Physics, Las Vegas, USA, 08–09 November 2017, Abstracts at, https://ato mic-nuclearphysics.conferenceseries.com/eposter/dozy-chaos-nature-of-molec ular-quantum-transitions-and-theory-of-the-optical-band-shapes-in-polymethinedyes-atomic-physics-2017, https://www.omicsonline.org/proceedings/dozychaos -nature-of-molecular-quantum-transitions-and-theory-of-the-optical-band-shapes -in-polymethine-dyes-79484.html.
- [36] V.V. Egorov, Where and why quantum mechanics ceases to work in molecular and chemical physics, in: European XFEL Theory Seminar, Schenefeld, Hamburg, Germany, 6 March, 2018. Abstract at, https://indico.desy.de/indico/event/ 20069/.
- [37] E.E. Jelley, Spectral absorption and fluorescence of dyes in the molecular state, Nature (London) 138 (1936) 1009–1010.
- [**38**] E.E. Jelley, Molecular, nematic and crystal states of 1: 1'-diethyl-ψ-cyanine chloride, Nature (London) 139 (1937) 631–632.
- [39] G. Scheibe, Über die Veränderlichkeit des Absorptionsspektrums einiger Sensibilisierungsfarbstaffe und deren Ursache (Variability of the Absorption Spectra of some Sensitizing Dyes and its Cause), Angew. Chem. 49 (1936) 563.
- [40] G. Scheibe, Ueber die Veraenderlichkeit der Absorptionsspektren in Loesungen und die Nebenvalenzen als ihre Ursache (On the variability of the absorption spectra in solutions and the secondary bonds as its cause), Angew. Chem. 50 (1937) 212–219.
- [41] L.G.S. Brooker, R.H. Sprague, C.P. Smith, G.L. Lewis, Color and constitution. I. Halochromism of anhydronium bases related to the cyanine dyes, J. Am. Chem. Soc. 62 (1940) 1116–1125.
- [42] A.H. Herz, Aggregation of sensitizing dyes in solution and their adsorption onto silver halides, Adv. Colloid Interface Sci. 8 (1977) 237–298.
- [43] T.H. James (Ed.), The Theory of the Photographic Process, Macmillan, New York, NY, 1977.
- [44] O.D. Kachkovski, O.I. Tolmachov, Yu.L. Slominskii, M.O. Kudinova, N.O. Derevyanko, O.O. Zhukova, Electronic properties of polymethine systems 7: soliton symmetry breaking and spectral features of dyes with a long polymethine chain, Dyes Pigments 64 (2005) 207–216.
- [45] V.V. Egorov, Effects of fluctuations in the transparency of the barrier in proton transfer reactions, Russ. J. Phys. Chem. 64 (1990) 1245–1254. https://www .researchgate.net/publication/286450441_Effects_of_fluctuations_in_the_transpar ency_of_the_barrier_in_proton_transfer_reactions. https://www.researchgate.net/ publication/287998475_V_V_Egorov_Zhurnal_Fizicheskoj_Khimii_64_2305-2324_1990.
- [46] K. Naito, A. Miura, Photogenerated charge storage in hetero-Langmuir-Blodgett films, J. Am. Chem. Soc. 115 (1993) 5185–5192.
- [47] J.N. Brönsted, Acid and basic catalysis, Chem. Rev. 5 (1928) 231-338.
- [48] V.V. Egorov, Where and why quantum mechanics ceases to work in molecular physics, Res. Gate (2018). (Accessed 8 January 2018).
- [49] V.V. Egorov, Explanation for "pedestrians" why quantum mechanics ceases to work in molecular physics, Res. Gate (2018). (Accessed 26 February 2018).
- [50] S. Dähne, Color and constitution: one hundred years of research, Science 199 (1978) 1163–1167.

- [51] A.D. Kachkovskii, The nature of electronic transitions in linear conjugated systems, Russ. Chem. Rev. 66 (1997) 647–664.
- [52] V.V. Egorov, Theory of tunnel transfer, Khim. Fiz. 7 (1988) 1466–1482 [Sov. J. Chem. Phys. 7, 1991, 2629–2660], https://www.researchgate.net/publication/28 7998398_V_V_Egorov_Theory_of_tunnel_transfer_Khim_Fiz_1988_7_1466-1482_ Sov_J_Chem_Phys_1991_7_2629-2660.
- [53] M. Born, J.R. Oppenheimer, Quantum theory of the molecules, Ann. Phys. (Leipzig) 84 (1927) 457–484.
- [54] J. Franck, E.G. Dymond, Elementary processes of photochemical reactions, Trans. Faraday Soc. 21 (1925) 536–542.
- [55] E.U. Condon, A theory of intensity distribution in band systems, Phys. Rev. 28 (1926) 1182–1201.
- [56] E.U. Condon, Nuclear motions associated with electron transitions in diatomic molecules, Phys. Rev. 32 (1928) 858–872.
- [57] E.U. Condon, The Franck-Condon principle and related topics, Am. J. Phys. 15 (1947) 365–374.
- [58] W.E. Lamb, R.C. Retherford, Fine structure of the hydrogen atom by a microwave method, Phys. Rev. 72 (1947) 241–243.
- [59] H.A. Bethe, The electromagnetic shift of energy levels, Phys. Rev. 72 (1947) 339–341.
- [60] P. Ehrenfest, Adiabatic invariants and quantum theory, Ann. Phys. (Leipzig) 51 (1916) 327–352.
- [61] A.S. Davydov, Quantum Mechanics, Pergamon Press, Oxford, UK, 1976.
- [62] G. Herzberg, J.W.T. Spinks, Molecular Spectra and Molecular Structure. 1. Spectra of Diatomic Molecules, Prentice-Hall, New York, 1939.
- [63] G. Herzberg, Molecular Spectra and Molecular Structure. 2. Infrared and Raman Spectra, D. Van Nostrand, Princeton, 1945.
- [64] G. Herzberg, Molecular Spectra and Molecular Structure. 3. Electronic Spectra and Electronic Structure of Polyatomic Molecules, Van Nostrand Reinhold, New York, London, 1966.
- [65] M.D. Frank-Kamenetskii, A.V. Lukashin, Electron-vibrational interactions in polyatomic molecules, Sov. Phys. Uspekhi 18 (1975) 391–409.
- [66] I.B. Bersuker, V.Z. Polinger, Vibronic Interactions in Molecules and Crystals, Springer, New York, 1989.
- [67] M. Stanke, Adiabatic, Born-Oppenheimer, and non-adiabatic approaches, in: J. Leszczynski, et al. (Eds.), Handbook of Computational Chemistry, Springer, Cham, Switzerland, 2017, pp. 173–223.
- [68] I.B. Bersuker, The Jahn-Teller Effect, Cambridge University Press, Cambridge, UK, 2006.
- [69] I.B. Bersuker, Pseudo-Jahn-Teller effect a two-state paradigm in formation, deformation, and transformation of molecular systems and solids, Chem. Rev. 113 (2013) 1351–1390.
- [70] Structures and conformations of non-rigid molecules, in: J. Laane, M. Dakkouri, B. van der Veken, H. Oberhammer (Eds.), NATO ASI Series (Series C: Mathematical and Physical Sciences), vol. 410, Springer, Dordrecht, Netherlands, 1993.
- [71] Structure and dynamics of non-rigid molecular systems, in: Y.G. Smeyers (Ed.), Topics in Molecular Organization and Engineering, vol. 12, Springer, Dordrecht, Netherlands, 1995.
- [72] Yu.E. Perlin, Modern methods in the theory of many-phonon processes, Sov. Phys. Uspekhi 6 (1964) 542–565.
- [73] S.I. Pekar, To the theory of luminescence and light absorption by impurities in dielectrics, Zh. Eksp. Teor. Fiz. 22 (1952) 641–657 (in Russian).
- [74] S.I. Pekar, On the effect of lattice deformations by electrons on optical and electrical properties of crystals, Usp. Fiz. Nauk 50 (1953) 197–252 (in Russian).
- [75] V.V. Egorov, The method of generating polynomials in the theory of electron transfer, in: Proc. Of the Second All-Union Conf. On Quantum Chemistry of Solid State, Lielupe, LatvSSR, 8–11 October 1985, p. 141 (in Russian).
- [76] M.A. Krivoglaz, S.I. Pekar, The shape of the spectra of the impurity light absorption and luminescence in dielectrics, Tr. Inst. Fiz. Akad. Nauk UKR SSR 4 (1953) 37–70 (in Russian).
- [77] M.A. Krivoglaz, The theory of thermal transitions, Zh. Eksp. Teor. Fiz. 25 (1953) 191–207 (in Russian).
- [78] E. Schrödinger, Statistical Thermodynamics, Cambridge University Press, Cambridge, UK, 1946.
- [79] V.V. Egorov, Electrodynamics of extended multiphonon transitions, Thesis for Doct. Phys-Math. Sci. Degree, in: Russian; Russ. Acad. Sci., Inst. Phys. Chem., Moscow, 2004, 255 pp.
- [80] A.I. Baz', Ya.B. Zel'dovich, A.M. Perelomov, Rasseyanie, Reaktsii I Raspady V Nerelyativistskoi Kvantovoi Mekhanike (Scattering, Reactions, and Decays in Nonrelativistic Quantum Mechanics), second ed., Nauka, Moscow, 1971 (in Russian).
- [81] E. Fermi, Sul moto dei neutroni nelle sostanze idrogenate (On the motion of neutrons in hydrogenated substances), Ricerca Sci. Ser. II 7 (1–2) (1936) 13–52.
- [82] Yu.N. Demkov, V.N. Ostrovskii, Method of Zero-Radius Potential in Atomic Physics, LGU, Leningrad, 1975 (in Russian).
- [83] F.I. Dalidchick, G.K. Ivanov, Electron in a periodic chain with impurities, Theor. Exp. Chem. 7 (1971) 147–153 (in Russian).
- [84] O. Sil'd, Zero-Phonon Lines and Spectral Hole Burning in Spectroscopy and Photochemistry, Springer-Verlag, Berlin, 1988.
- [85] M. Abramowitz, I.A. Stegun, Handbook of Mathematical Functions, Dover P.I., New York, 1965.
- [86] S.I. Pekar, Theory of F-centers, Zh. Eksp. Teor. Fiz. 20 (1950) 510–522 (in Russian).
- [87] R.A. Marcus, On the theory of oxidation-reduction reactions involving electron transfer, J. Chem. Phys. 24 (1956) 966–978.

V.V. Egorov

- [88] R.A. Marcus, Electrostatic free energy and other properties of states having nonequilibrium polarization, J. Chem. Phys. 24 (1956) 979–989.
- [89] R.A. Marcus, On the theory of oxidation-reduction reactions involving electron transfer. II. Applications to data on the rates of isotopic exchange reactions, J. Chem. Phys. 26 (1957) 867–871.
- [90] R.A. Marcus, On the theory of oxidation-reduction reactions involving electron transfer. III. Applications to data on the rates of organic redox reactions, J. Chem. Phys. 26 (1957) 872–877.
- [91] R.A. Marcus, Norman Sutin, Electron transfers in chemistry and biology, Biochim. Biophys. Acta 811 (1985) 265–322.
- [92] R.A. Marcus, Electron transfer reactions in chemistry. Theory and experiment, Rev. Mod. Phys. 65 (1993) 599–610.
- [93] K. Huang, A. Rhys, Theory of light absorption and non-radiative transitions in Fcentres, Proc. R. Soc. A 204 (1950) 406–423.
- [94] M. Lax, The Franck-Condon principle and its application to crystals, J. Chem. Phys. 20 (1952) 1752–1760.
- [95] M. Planck, On the law of distribution of energy in the normal spectrum, Ann. Phys. (Leipzig) 309 (1901) 553–563.
- [96] V.V. Egorov, Where and Why Quantum Mechanics Ceases to Work in Molecular and Chemical Physics, European XFEL Theory Seminar, Invited Talk, 6 March 2018.
- [97] M.W. Jarosik, R. Szczę śniak, A.P. Durajski, J.K. Kalaga, W. Leoński, Influence of external extrusion on stability of hydrogen molecule and its chaotic behavior, Chaos 28 (2018) 013126-1–013126-6.
- [98] V.F. Gantmakher, M.V. Feigel'man, Mesoscopic unification, Phys. Uspekhi 41 (1998) 105–108.
- [99] R.A. Jalabert, Mesoscopic transport and quantum chaos, Scholarpedia 11 (1) (2016) 30946.
- [100] O.V. Prezhdo, V.V. Kisil, Mixing quantum and classical mechanics, Phys. Rev. A 56 (1997) 162–176.
- [101] D.V. Makhov, W.J. Glover, T.J. Martinez, D.V. Shalashilin, *Ab initio* multiple cloning algorithm for quantum nonadiabatic molecular dynamics, J. Chem. Phys. 141 (2014) 054110-1–054110-11.
- [102] D.V. Zhdanov, D.I. Bondar, T. Seideman, Friction as a consistent quantummechanical concept, Phys. Rev. A 98 (2018) 042133-1–042133-7.
- [103] M.C. Gutzwiller, a) Chaos in Classical and Quantum Mechanics, Springer-Verlag, New York, 1990 and references therein;
 b) Scientific American 1, 1992, pp. 78–84.
- [104] M.V. Berry, The Bakerian lecture, 1987: quantum chaology, Proc. R. Soc. A 413 (1987) 183–198.
- [105] M.V. Berry, Quantum chaology, not quantum chaos, Phys. Scr. 40 (1989) 335–336.
- [106] E.J. Heller, Bound-state eigenfunctions of classically chaotic Hamiltonian systems: scars of periodic orbits, Phys. Rev. Lett. 53 (1984) 1515–1518.
- [107] B.V. Chirikov, Transient chaos in quantum and classical mechanics, Found. Phys. 16 (1986) 39–49.
- [108] H.J. Stöckmann, Quantum Chaos: an Introduction, Cambridge Univ. Press, Cambridge, 1999 and references therein.
- [109] M.C. Gutzwiller, Periodic orbits and classical quantization conditions, J. Math. Phys. 12 (1971) 343–358.
- [110] M.V. Berry, Quantizing a classically ergodic system: Sinai's billiard and the KKR method, Ann. Phys. (N.Y.) 131 (1981) 163–216.
- [111] M.V. Berry, Quantum scars of classical closed orbits in phase space, Proc. R. Soc. Lond. A 423 (1989) 219–231.
- [112] E.J. Heller, Mode mixing and chaos induced by potential surface crossings, J. Chem. Phys. 92 (1990) 1718–1727.
- [113] L. Kaplan, E.J. Heller, Linear and nonlinear theory of eigenfunction scars, Ann. Phys. (N. Y.) 264 (1998) 171–206.

- [114] H. Fujisaki, K. Takatsuka, Chaos induced by quantum effect due to breakdown of the Born-Oppenheimer adiabaticity, Phys. Rev. E 63 (2001) 066221-1-066221-10.
- [115] H. Higuchi, K. Takatsuka, Quantum chaos induced by nonadiabatic coupling in wave-packet dynamics, Phys. Rev. E 66 (2002) 035203-1–035203-4.
- [116] H. Fujisaki, Quantum-"classical" correspondence in a nonadiabatic transition system, Phys. Rev. E 69 (2004) 037201-1–037201-4.
- [117] Yu. Dabaghian, R. Jensen, Quantum chaos in elementary quantum mechanics, Eur. J. Phys. 26 (2005) 423–439.
- [118] A. Wasserman, N.T. Maitra, E.J. Heller, Investigating interaction-induced chaos using time-dependent density-functional theory, Phys. Rev. A 77 (2008) 042503-1–042503-7.
- [119] M. Fujii, K. Yamashita, Semiclassical quantization of nonadiabatic systems with hopping periodic orbits, J. Chem. Phys. 142 (2015) 074104-1–074104-10.
- [120] B.V. Chirikov, A universal instability of many-dimensional oscillator systems, Phys. Rep. 52 (1979) 263–379, and references therein.
- [121] A. Loskutov, Fascination of chaos, Phys. Uspekhi 53 (2010) 1257-1280.
- [122] S.P. Kuznetsov, Dynamical chaos and uniformly hyperbolic attractors: from mathematics to physics, Phys. Uspekhi 54 (2011) 119–144.
- [123] S.P. Kuznetsov, Hyperbolic Chaos: A Physicist's View, Springer-Verlag, Berlin, Heidelberg, 2012.
- [124] R.P. Feynman, An operator calculus having applications in quantum electrodynamics, Phys. Rev. 84 (1951) 108–128.
- [125] R. Kubo, Thermal ionization of trapped electrons, Phys. Rev. 86 (1952) 929–937.[126] R. Kubo, Y. Toyozawa, Application of the method of generating Function to
- [120] K. Kubo, T. Hydzawa, Application of the method of generating Function to radiative and non-radiative transitions of a trapped electron in a crystal, Prog. Theor. Phys. 13 (1955) 160–182.
- [127] R.P. Feynman, A.R. Hibbs, Quantum Mechanics and Path Integrals, McGraw-Hill, New York, 1965.
- [128] B.B. Kadomtsev, Dynamics and information, Phys. Uspekhi 37 (1994) 425-499.
- [129] S.Ya. Kilin, Quantum information, Phys. Uspekhi 42 (1999) 435-452.
- [130] M.B. Mensky, Quantum mechanics: new experiments, new applications, and new formulations of old questions, Phys. Uspekhi 43 (2000) 585–600.
- [131] K.A. Valiev, Quantum computers and quantum computations, Phys. Uspekhi 48 (2005) 1–36.
- [132] A.L. Fradkov, Application of cybernetic methods in physics, Phys. Uspekhi 48 (2005) 103–127.
- [133] M. Paternostro, On the mechanism for optomechanical entanglement and its revelation, J. Phys. B At. Mol. Opt. Phys. 41 (2008) 155503-1–155503-6.
 [134] W.H. Zurek, Quantum darwinism, Nat. Phys. 5 (2009) 181–188.
- [135] L. Mazzola, M. Paternostro, Activating optomechanical entanglement, Nat. Sci. Rep. 1 (2011) 199-1–199-5.
- [136] A. Kronwald, F. Marquardt, A.A. Clerk, Arbitrarily large steady-state bosonic squeezing via dissipation, Phys. Rev. A 88 (2013) 063833-1-063833-10.
- [137] A. Kronwald, F. Marquardt, A.A. Clerk, Dissipative optomechanical squeezing of light, New J. Phys. 16 (2014) 063058-1–063058-21.
- [138] D. Wigger, S. Lüker, V.M. Axt, D.E. Reiter, T. Kuhn, Squeezed phonon wave packet generation by optical manipulation of a quantum dot, Photonics 2 (2015) 214–227.
- [139] J. Tuziemski, J.K. Korbicz, Objectivisation in simplified quantum brownian motion models, Photonics 2 (2015) 228–240.
- [140] M.G. Genoni, M. Bina, S. Olivares, G. De Chiara, M. Paternostro, Squeezing of mechanical motion via qubit-assisted control, New J. Phys. 17 (2015) 013034-1–013034-9.
- [141] C. Pellegrini, A. Marinelli, S. Reiche, The physics of x-ray free-electron lasers, Rev. Mod. Phys. 88 (2016) 015006-1–015006-55.
- [142] V.V. Egorov, Quantum-classical mechanics: luminescence spectra in polymethine dyes and J-aggregates. Nature of the small Stokes shift, Res. Phys. 13 (2019) 102252-1–102252-14.